Montclair State University

Montclair State University Digital Commons

Theses, Dissertations and Culminating Projects

5-2020

Tag-Team Back Again: Using Memory as Method to Understand the Intergenerational Transmission of Egalitarian Parenting in My **Black Family**

Rebecca M. Swann

Follow this and additional works at: https://digitalcommons.montclair.edu/etd



TAG-TEAM BACK AGAIN:

USING MEMORY AS METHOD TO UNDERSTAND THE INTERGENERATIONAL TRANSMISSION OF EGALITARIAN PARENTING IN MY BLACK FAMILY

A DISSERTATION

Submitted to the Faculty of

Montclair State University in partial fulfillment

of the requirements

for the degree of Doctor of Philosophy

by

REBECCA M. SWANN

Montclair State University

Upper Montclair, NJ

May 2020

Dissertation Chair: Dr. Jennifer Urban

MONTCLAIR STATE UNIVERSITY

THE GRADUATE SCHOOL

DISSERTATION APPROVAL

We hereby approve the Dissertation

TAG-TEAM BACK AGAIN: USING MEMORY AS

METHOD TO UNDERSTAND THE

INTERGENERATIONAL TRANSMISSION OF

EGALITARIAN PARENTING IN MY BLACK FAMILY

of

Rebecca M. Swann

Candidate for the Degree:

Doctor of Philosophy

	Dissertation Committee:
Graduate Program:	
Family Science and Human Development	
Certified by:	Dr. Jennifer Urban
	Dissertation Chair
	Dr. Kathryn Herr
Dr.M. Scott Herness	
Vice Provost for Research	
and Dean of the Graduate School	
	Dr. Bradley van-Eeden-Moorefield
April 27 2020	
April 27, 2020	D 161 111
Date -	Dr. Michael Hannon



ABSTRACT

TAG-TEAM BACK AGAIN:

USING MEMORY AS METHOD TO UNDERSTAND THE INTERGENERATIONAL TRANSMISSION OF EGALITARIAN PARENTING IN MY BLACK FAMILY

by Rebecca M. Swann

Life course theory's (Elder, 1998) principles of *linked lives* and *historical time and place* can be used to understand how attitudes, values, and behaviors are passed down across generations amid the historical context of Black families in the United States. This dissertation used autoethnography to explore the construction and transmission of egalitarianism, allowing the researcher to be both participant and analyst. Qualitative data consisted of critical reflections and 17 individual oral history interviews with family members across four generations. Procedures outlined in Gilligan's Listening Guide were used to analyze data, resulting in pronoun-poems for each interviewee and generation. Individual voices and generational experiences are highlighted in poems. Data were analyzed within and across individuals to identify generational constructions of egalitarianism, including how these constructs were adapted as they transferred from generation to generation amid varying societal contexts. This study produced three major themes: parental socialization, partnering socialization, and intersectional socialization. Findings from this study suggest that Black families share responsibilities to achieve the common good, with respect and attention to the needs of individual family members.

Keywords: autoethnography, oral history, egalitarianism, parenting, life course theory, Black family, intergenerational, Gilligan's Listening Guide

ACKNOWLEDGEMENT

All my life, I've listened to the stories of my ancestors. I could never grow tired of hearing a tale one more time. Some stories may never make it to a wider audience, but they are no less important. I am motivated to share the untold stories, bringing unsung heroes to the fore and giving voice to the voiceless. With this whisper, I have endeavored to bring all our voices forward.

I am privileged to have been supported by teams of people who were willing to puzzle with me, strategize methodology, help me figure out how to amplify my story, send an encouraging text, motivate me to keep going, check-in on my mental health, and offer solidarity. You compassionate souls have kept me moving toward the finish. I could not possibly list you all, but you know who you are. I see you. I honor you. I appreciate you. I love you.

Aunt Sue has a head full of stories.
Aunt Sue has a whole heart full of stories.
Summer nights on the front porch
Aunt Sue cuddles a brown-faced child to her bosom
And tells him stories.

...And the dark-faced child, listening, Knows that Aunt Sue's stories are real stories. He knows that Aunt Sue never got her stories Out of any book at all, But that they came Right out of her own life.

The dark-faced child is quiet Of a summer night Listening to Aunt Sue's stories.

Excerpts from *Aunt Sue's Stories* (Hughes, 1926, p. 40)

DEDICATION

This work is dedicated to my family and friends, those with us in a physical form and those in spirit. None of this would have been possible without your constant encouragement, support, and willingness to rearrange your lives to give me space to dream and write. Thank you for sharing your stories with me. I am forever grateful.

Grandma Jeanne – You saw me as a Black feminist before I knew I would be and opened my eyes to Toni, Zora, Alice, and Ntozake. Your subtle influence helped shape my life.

Grandma Priscilla – They say, "You cannot be who you cannot see." I saw you. You have always been the inspiration.

Grandpa Joe - You were the invisible glue that held us together. Thank you.

Old Pop – Your crazy tales always had a grain of truth. I still hear your stories in my head.

Daddy – I struggle to find the words, so I will say: 'I still hang around, neither lost nor found. I hear the lonely sound of music in the night.' You are loved. You are missed...all day, every day.

Mom – Your belief in me always has carried me through. You are my lighthouse. Thank you for your unconditional love and ready ear.

Marcus and Drew – You two are my greatest accomplishments. Thank you for your love, patience, and understanding. I look forward to your stories most of all.

Damian, my life partner – You are everything I never knew I needed. This is for us.

TABLE OF CONTENTS

Error! Bookmark not defined.	CHAPTER 1: INTRODUCTION
Error! Bookmark not defined.	CHAPTER 2: REVIEW OF LITERATURE
Error! Bookmark not defined.	CHAPTER 3: METHODS
Error! Bookmark not defined.	CHAPTER 4: FINDINGS
80	CHAPTER 5: DISCUSSION
Error! Bookmark not defined.94	REFERENCES
Error! Bookmark not defined.9	APPENDIX A: IN-PERSON AND TELEPHONE PLEA
Error! Bookmark not defined.	APPENDIX B: INTERVIEW PROTOCOLS
Error! Bookmark not defined.9	APPENDIX C: ADULT CONSENT FORM
Error! Bookmark not defined.	APPENDIX D: PARENT/GUARDIAN CONSENT FORM
Error! Bookmark not defined.35	APPENDIX E: ASSENT FORM

PREFACE

how to listen #1

Somewhere in my brain each laugh, tear and lullaby becomes *memory*.

writing #1

It's easier to make up stories than it is to write them down. When I speak, the words come pouring out of me. The story wakes up and walks all over the room. Sits in a chair, crosses one leg over the other, says, Let me introduce myself. Then just starts going on and on. But as I bend over my composition notebook, only my name comes quickly. Each letter, neatly printed between the pale blue lines. Then white space and air and me wondering, How do I spell introduce? Trying again and again until there is nothing but pink bits of eraser and a hole now where a story should be.

how to listen #7

Even the silence has a story to tell you. Just listen. Listen.

how to listen #10

Write down what I think I know. The knowing will come.

Just keep listening ...

- Excerpts from *Brown Girl Dreaming* (Woodson, 2014)

CHAPTER 1: INTRODUCTION

There is a photo of my 37-year old grandmother, Priscilla Sisco Swann in front of the White House. The photo was taken by my grandfather; he drove her there to attend National Head Start Day hosted by First Lady Ladybird Johnson. It was 1965. My grandmother was a founder of a Head Start program back home in New York State. She also worked part-time and was raising a family with my grandfather. My grandparents had six sons. At the time this photo was taken, my father, the eldest, was 18 and my uncle was five. I have always been inspired by this photo. For me, it represents egalitarian values and shared responsibilities between spouses. It represents flexible gender roles and support toward common goals. It represents the tenacity and determination of women, particularly Black women. This photo is the heart of this study.



Figure 1. Priscilla Sisco Swann standing in front of the White House

Framed by life course theory, this autoethnographic study explores my family's construction of egalitarianism and its transmission over four generations. The purpose of this study is to understand how egalitarianism was constructed and passed down. The processes and practices of families like ours are not accounted for in the family science literature (Trask, 2018; van Eeden-Moorefield & Shih, 2015) and are particularly important for individuals and families who are not commonly participants in studies of egalitarian parenting. I hope to contribute to conversations in family science about intergenerational transmission of egalitarian Black family

processes. This study required me to critically reflect on my experiences as a Black granddaughter, daughter, niece, sister, cousin, wife, and mother. The data used for this study includes my reflections along with the memories and narratives shared by 14 members of my family. Throughout this dissertation, I provide personal references, recollections, and facts based on familial ancestral research to keep the reader grounded in my family's experiences. Humbly, I request that the reader allow my use of scholarly personal narrative (Nash, 2004), which is an approach to telling one's own story. In this study, I integrate family theory, social-historical context, and research methodology to build a multivocal narrative of how egalitarianism is constructed and transmitted in my Black family. Autoethnography "shows struggle, passion, embodied life, and the collaborative creation of sense-making" and "was designed to be unruly, dangerous, vulnerable, rebellious, and creative" (Ellis & Bochner, 2006, p. 433). When integrating these three concepts (i.e., theory, context, and methodology) this dissertation frequently uses a familiar tone, a first-person voice, and non-standard academic style.

Problem Statement

One of the greatest challenges for dual-earner families is maintaining work-family balance. Two incomes are necessary to afford most families' current standard of living (Heckert, Nowak, & Snyder, 1998; Gerson, 2010). Many families with two working parents negotiate responsibilities in order to feel they are successful caretakers of their families and careers. Often, working parents use egalitarianism to share responsibilities for earning wages, maintaining the household, and caring for children. Studies done at the time of the Women's Liberation movements (Beckett and Smith, 1981; Dietrich, 1975; Scanzoni, 1975; Willie & Greenblett, 1978) showed that Black families were more likely than White families to exhibit egalitarianism. In 1978, married Black women had a greater employment rate (noted by labor force

participation) than married White women (58% and 47%, respectively) continuing and narrowing a trend that began in 1890 (22% and 2%, respectively; Goldin, 1977). Since that time however, there has been a dearth of research on dual-income Black families in general and there have been no studies on how Black families construct egalitarian processes. There have been cross-race studies (Beckett & Smith, 1981; Dietrich, 1975; Dugger, 1988; Glauber & Gozjolko, 2011; Raley, Mattingly, & Bianchi, 2006; Scanzoni, 1975; Willie & Greenblatt, 1978) on the degree to which people have egalitarian gender attitudes.

Individuals in families exhibited more egalitarian gender attitudes in the 1970s and 1980s than they ever had before (Carr, 2004; Cotter, Hermsen, & Vanneman, 2011; Min, Silverstein, & Lendon, 2012; Moen, Erickson, & Dempster-McClain, 1997). However, attitudes do not always translate to behavior. Despite this, more Black than White fathers are involved in caregiving interactions with their children (Jones & Mosher, 2013). Moreover, Black men and women have been part of the labor force for hundreds of years, whether paid or unpaid. As a result, it may be assumed that Black couples have had more experience negotiating how to work together to share responsibilities for maintaining a home and raising children. Egalitarianism appears to be manifested in both the attitudes and behaviors of Black families (Becker & Moen, 1999; Beckett & Smith, 1981; Curenton, Crowley, & Mouzon, 2018; Dietrich, 1975; Jones & Mosher, 2013; Mack, 1978; Middleton & Putney, 1960; Perry, Harmon, & Leeper, 2012; Raley, Mattingly, & Bianchi, 2006; Scanzoni, 1975; Thornton & Young-DeMarco, 2001; White, 2006a, 2006b; Willie & Greenblatt, 1978).

There is evidence that Black families exhibit egalitarianism when looking at point-intime studies but there has not been a study that looks at the development of egalitarianism across generations. Life course theory and its principles of *linked lives* and *historical time and place* can be used to explain the intergenerational transmission of egalitarianism in the United States from the 1960s to the present. Life course principles theorize that families pass down egalitarian attitudes and behaviors to younger generations. Trask (2018) underscores the importance of using life course theory to study how race, ethnicity, and cultural diversity affect individual and family development, especially the critical analyses of context, time, and process. Further, the increase of women in the workforce has rearranged family processes whereby traditionally patriarchal values are being substituted with more egalitarian processes (Trask, 2018). Life course theory and these specific principles can provide a framework for the study of how Black families in the United States construct and transfer egalitarianism to their descendants, but first the way Black families have been portrayed in family science scholarship needs to be addressed.

There also is a problem in how scholars have written about Black families. Western studies of families have often focused on heterosexual, two-parent households (often White) perpetuating a deficit-perspective of families who did not meet these criteria (Coontz, 2011; Trask, 2018; van Eeden-Moorefield & Shih, 2015). Further, most research on dual-earner couples includes only White participants (Becker & Moen, 1999; Carr, 2004; Cotter, Hermsen, & Vanneman, 2011; Daly, 2001; Min, Silverstein, & Lendon, 2012; Moen, Erickson, & Dempster-McClain, 1997). The lack of inclusion of different races in these studies leads family scholars to identify White families and their associated values and priorities as the norm. Further, research on Black families often has focused on the pathological aspects of Black parenting. Many studies have examined the failures or pathologies of Black families, including absent, nonresident, incarcerated, or unmarried parents (Thomas, Farrell, & Barnes, 1996) and matriarchical family structures (Frazier, 1948; Moynihan, 1965). Other scholars have attempted to provide more strengths-based accounts of Black fathering (Cochran, 1997; Furstenberg, 2007;

5

Gadsden & Smith, 1994; Hamer, 1997; McAdoo & McAdoo, 1997). The problem of focusing on Black family pathology highlights the cultural deficit perspective through which they are viewed. The benchmark "Standard North American Family," or SNAF (Coontz, 2011; van Eeden-Moorefield & Shih, 2015), consists of a male breadwinner and female homemaker who are White and middle-class. The SNAF family has been used as a family reference point against which all other family types are compared. A strengths-based account of Black family process that explores how Black families construct egalitarianism and pass down this value over generations would advance the field by adding the complexities of diverse families. The stories Black families tell about resilience and strength need to be collected and shared to inform a narrative of how Black families operate. Interventions with Black families also should consider an increased emphasis on ancestors (extended family) and the role they play in family process. A scholarly personal narrative, like this study, can provide scholars, and other consumers of family science, evidence of a Black family that has collaborated overtime to achieve a collective aim — the societal advancement of women.

In my family, egalitarianism looks like shared humanity and responsibility across gender lines, within marriage or between co-parents. You can see examples of this cooperation in the way families communicate, establish shared goals, and support one another toward these goals, including males accepting family roles that challenge hegemonic expectations of men. This means that fathers provide nurturing care for children and perform household responsibilities in order to allow mothers to pursue academic and career goals. In their stories of partnership, my family members recognized the value of community and emphasized familial harmony and compassion over generations.

Theoretical and Conceptual Framework

This study is framed by life course theory's principals of *linked lives* and *historical time* and place, concerning how values, attitudes, and behaviors are passed down over generations. It also includes the historical context of Black families in the United States as well as concepts of egalitarianism and work-family balance.

Life course theory explains individual development within time and space and the degree to which lives change in relation to context. The principle of *linked lives* often refers to the multiple generations in a family through which values, attitudes, and behaviors are passed down. The principle of historical time and place refers to the social-historical context of these intergenerational relationships. Over generations, the Black family has endured a legacy of oppression during slavery, Jim Crow, and currently mass incarceration. Many Black couples have been dual-wage earners since the Post-Reconstruction period and may have developed egalitarianism, a type of cooperative decision-making and role sharing that emphasizes gender neutrality and the care of both work and family. Over generations, Black men and women have shown egalitarian attitudes and behaviors (Becker & Moen, 1999; Beckett & Smith, 1981; Curenton, Crowley, & Mouzon, 2018; Dietrich, 1975; Glauber & Gozjolko, 2011; Jones & Mosher, 2013; Kane, 2000; Mack, 1978; Middleton & Putney, 1960; Orbuch & Eyster, 1997; Perry, Harmon, & Leeper, 2012; Raley, Mattingly, & Bianchi, 2006; Scanzoni, 1975; Thornton & Young-DeMarco, 2001; White, 2006a, 2006b; Willie & Greenblatt, 1978). Cross-race studies have shown stronger egalitarian attitudes and behaviors among Black husbands as compared to White husbands (Glauber & Gozjolko, 2011). My study investigated how my conception and practice of egalitarian parenting was influenced by experiences of previous generations of my

family as well as the social-historical context in which members of my family grew up (the United States from the 1930's through the 1980's)

Methodology

I used autoethnography to study how egalitarianism was constructed and passed down in my Black family. This research method allowed me to explore and more deeply understand the culture of egalitarianism in my family. Autoethography combines autobiographical and personal storytelling with cultural and social contexts (Boylorn & Orbe, 2014) and emphasizes reflection and positionality. Autoethnography provided me the opportunity to study my natural settings and make sense of and interpret egalitarianism as it was passed down to me (Denzin & Lincoln, 2000, p. 3) while recognizing the social-historical context. This autoethnography included critical reflection, journaling, and in-depth individual oral history interviews with 13 family members across four generations. In addition, I reviewed old photographs and historical documents. Procedures outlined in Gilligan's Listening Guide (2015) were used to analyze qualitative data.

Significance

This study is significant to the field of family science because it brings attention to a contribution made by an understudied group. Though many Black couples have been dual-wage earners since the Post-Reconstruction period, they have been understudied in research on workfamily balance. "Negative spaces are the recessive areas that we are unaccustomed to seeing but that are every bit as important for the representation of the reality at hand" (Daly, 2003, p. 771). Black families' egalitarian attitudes and behaviors are the "negative spaces" in the study of work-family balance.

Organization

This dissertation is organized into chapters. Chapter 1 presents the introduction, Chapter 2 provides a review of theory and relevant literature and identifies the research question. Chapter 3 describes the methodology used. Chapter 4 presents the findings. Finally, Chapter 5 discusses the findings in conversation with the broader literature.

CHAPTER 2: REVIEW OF LITERATURE

This chapter outlines the theoretical and conceptual framework of this study. It begins with a presentation of life course theory and discussion of the principles of *linked lives* and *historical time and place*, continues with a presentation of the historical context of Black families in the United States, and concludes with a discussion of scholarship on dual-earner families and the negotiation of responsibilities in a household. All of these concepts are important to understanding the intergenerational transmission of egalitarianism in Black families.

Life course theory uses positional and temporal concepts to situate an individual within time and space. Social-historical context is of particular importance to understanding how lives change in relation to the external societal environment. This dissertation focuses on two principles of life course theory. Linked lives often refers to the multiple generations in a family. These intergenerational relationships are a key forum for the transmission of values, attitudes, and behaviors. Moreover, the social-historical context of such transmission may influence the extent to which the transmitted value proliferates or dwindles. During the 1970s and 1980s, egalitarian gender role attitudes increased and women entered the workforce perhaps as a result of social movements such as Civil Rights, Vietnam protests, and Women's Liberation, where young adults were empowered to raise their voices in opposition to the status quo. Historical data on employment indicates, however, that Black women were part of the workforce as early as 1890, participating at rates far exceeding that of White women (Goldin, 1977, McKay, 2007). Moreover, Black couples have demonstrated egalitarian attitudes about women in the workforce and exhibited egalitarian behaviors in the maintenance of their households for a long time (Beckett & Smith, 1981; Dietrich, 1975; Scanzoni, 1975; Willie & Greenblett, 1978). However, scholarship on the intergenerational transmission of gender role attitudes primarily focuses on

White families and these studies are most frequently quantitative, including the analyses of secondary data from large national, sometimes longitudinal, panel studies (Acock & Bengtson, 1978; Blee & Tickamyer, 1995; Cotter, Hermsen, & Vanneman, 2011; Cunningham, 2001; Glass, Bengtson, & Dunham, 1986; Min, Silverstein, & Lendon, 2012; Moen, Erickson, & Dempster-McClain, 1997; Platt & Polavieja, 2016; Roest, Dubas, & Gerris, 2010; Schofield & Abraham, 2017; Thornton, Alwin, & Camburn, 1983). This study examines how egalitarian behaviors, like parenting, are transmitted across generations and is framed by life course theory.

Theoretical Framework

Life Course Theory

Elder's life course theory (1998) posits that lives change in response to changing contexts. Society, biology, psychology, and family interact to shape an individual's development. These interactions may include war, economic crises, disease, political movements, stress, poverty, affluence, neighborhood composition, family structure and household membership, employment, and immigration. Context impacts lives. Ontogenetic theorists (e.g., Erikson) view the social world as a "scene or setting" through which individuals must pass. Conversely, life course theorists view the social world as a contributor to an individual's development. Relational development systems (RDS) theory sees this as a bi-directional process (Ford & Lerner, 1992; Overton, 2015). RDS is a meta-theory and Life Course theory is a mid-range theory that is included under the umbrella of RDS meta-theory. The individual influences and is influenced by the context.

A positional and temporal perspective guides life course theory. The theory's key terms make clear the linear movement of someone from one developmental state to another within a normative or age-graded span of time, including transition, turning point, trajectory, sequence,

timing, and age. *Transitions* refer to choice points or discrete events in a relatively short time period that result in a change of role or require additional responsibility. A *turning point* refers to a dramatic change in one's life (i.e., incarceration, reentry after incarceration). A *trajectory* is a plane that may "knife-off" from a previous state; it represents a change in direction. *Sequence* implies an understanding that there is a hierarchy or progression of development while timing is defined within the context of one's cohort. Finally, *age* is used to define birth cohorts. All individuals born within a particular span of time are expected to have experienced changes in society at the same developmental time point. Birth cohorts define what is "normal" for those in the cohort. Therefore, appropriate development is relative. These concepts are key to understanding the interplay between individuals and their social-historical context.

Life course theory can be visualized as an integration of three distinct clocks measuring ontogenetic time (individual biographies), generational time (cohort experiences) and, historical time (social and cultural contexts), respectively (Bengtson & Allen, 1993; Macmillan & Copher, 2005; Umberson, et al., 2010). The theory is guided by five principles (Elder, Shanahan, & Jennings, 2015): *life-span development, timing, human agency, historical time and place,* and *linked lives*.

The *principle of life-span development* emphasizes the study of individuals from birth to death; human development is a life-long process. This principle also concerns the temporal (i.e., age-based) cohorts or cohort-historical perspective (Elder, Shanahan, & Jennings, 2014). People in particular birth cohorts have distinct life experiences. *Cohort centrism* is the belief that a group of people born at the same historical time experience social change within a given culture in the same sequence and at the same age (e.g., Baby Boomers, Millennials). However, because of intrapersonal variability and the intersections of race, class and gender, one should not assume

that a cohort of people experience the world in the same way, otherwise known as a *cohort* fallacy.

The *principle of timing* indicates that the impact of context is contingent on the point in an individual's development at which an event occurs. Elder (1974/1999) integrated theories about developmental age and roles to study how timing, context, and agency affect development and socialization across the life course. In Elder's Great Depression study (1974/1999), the age of an individual during the worst of the economic crisis was crucial to understanding how context can impact development. For example, life for children who were young in the early 1930s was dependent on the status of their families while older children would have been expected to assist with financially supporting the family. The ages of children in this social-historical context are important to their developmental trajectories.

The *principal of human agency* refers to the choices made and actions taken by individuals within the opportunities and constraints of social-historical context. For example, children in Elder's Great Depression study (1974/1999) may have sought employment to provide economically for their families. Parents may have decided not to have more children to limit the number of individuals in their household.

The *principle of historical time and place* concerns the study of human lives in relation to social-historical context. Individuals are embedded in and shaped by the historical times and places they experience over their lifetime. They adapt their lives to accommodate changes in society. Elder's (1974/1999) study of youth growing up during the Great Depression connected youth development to the concepts of hardship, the importance of employment, and the timing of marriage.

Finally, the *principle of linked lives* positions the family as the primary arena for experiencing and interpreting the broader social world. The family and its network of interdependent, shared, and often reciprocal relationships link individual to environmental experiences. The developmental trajectories of other family members and social-historical context also influence an individual's developmental trajectory. In life course theory, individuals' lives are interconnected; social and historical context influences individuals through these interconnections. This principle explains how values, beliefs, traditions, and behaviors are passed down (or transmitted) from one generation to the next.

A Culturally-Responsive Life Course Theory

Life course theory is guided by temporal and positional concepts. Similarly, concepts of time and space also guide African religious and philosophical beliefs (Mbiti, 1969/1997). These concepts connect to the principle of *linked lives*. Adding African philosophy helps to make life course theory more culturally responsive because it focuses on the importance of intergenerational connections in Black families. African philosophy primarily is focused on two points of actual time— what was (past) and what is (present). Little consideration is given to what will be (future). In fact, in the Kikamba and Gikuyu languages, there are no verb tenses to describe behaviors in the distant future (beyond two years). Mbiti notes that, "A person experiences time partly in his individual life, and partly through the society which goes back many generations before his own birth" (1969/1997, p. 17). Though the future is limited to events within the next two years, the African duration of "past" is seemingly infinite.

The Swahili terms 'sasa' and 'zamani' are used to describe African concepts of time. Sasa refers to the immediate—things that are about to occur, in process, or recently experienced as well as all events experienced in the past. Sasa is an all-encompassing dimension, as it includes the present, limited future, and <u>experienced</u> past, and may be measured at the individual or community level. Mbiti (1969/1997) also refers to *sasa* as "micro-time." *Zamani* is an all-encompassing dimension, including the present, limited future, and the <u>infinite</u> past. However, Mbiti considers *Zamani* to be "macro-time", meaning that *Sasa* is incorporated into the *Zamani*. Mbiti (1969/1997) explains:

Sasa generally binds individuals and their immediate environment together. It is the period of conscious living. On the other hand, Zamani is the period of the myth, creating a sense of foundation or 'security' to the Sasa period; and binding together all created things, so that all things are embraced within the Macro Time. ... Such history and pre-history tend to be telescoped into a very compact, oral tradition and handed down from generation to generation... Man looks back from whence he came" (p. 22-23).

African religion and philosophy acknowledge that, over the life course, people travel from *Sasa* to *Zamani*. Even after death, people may exist in the *sasa* so long as there is someone who knew them personally and in name. They remain the 'living dead,' alive in the memory of those who knew them. My father, three of my grandparents, uncles and some cousins live in the *sasa*. More distant ancestors, who no one who is currently alive knew personally, are in the *zamani*. The *zamani* is where one can find the origins of traditions and customs. My great-great-grandparents, my enslaved ancestors, my immigrant great-grandfather, and ancestors beyond reside in the *zamani*. Mbiti (1969/1997) underscores the importance of family, knowledge of ancestors, and the role of oral history in the intergenerational transmission of values and behaviors. Both Elder's life course theory and African philosophy place importance on time, space, and process to explain how humans develop over their lives.

Life course theory's aforementioned principles of *linked lives* and *historical time and* place represent the intersection of individual biography and development (ontogenetic), generational experiences and trajectories, and social history. Individuals' experiences, customs, values, and attitudes are transmitted across generations within a family. The family is a "collection of individuals with shared history who interact within ever-changing social contexts across ever increasing time and space" (Bengtson & Allen 1993, p. 470). African philosophy notes that 'the family' extends into the *sasa* and *zamani* and includes the experiences of known and unknown ancestors. The integration of Elder's life course theory with Mbiti's conception of *sasa* and *zamani* allows scholars to make sense of the experiences of diverse families over generations (time) and contexts (space). Further, this integration helps to fend off the bias in family science studies (Trask, 2018; van-Eeden-Moorefield & Shih, 2015) to focus on Western conceptions of family.

Literature Review

Life Course Theory, Linked Lives, and Intergenerational Relationships

Intergenerational relationships within a family are critical to the life course principle of *linked lives*. Studies of multiple generations in a family have examined relationships between parents and children, and between grandparents and grandchildren (Ames, 1999; Greenfield & Marks, 2005; Hareven, 1994; Elder, 1974/1999; Umberson, Pudrovska, & Reczek, 2010). Calling the parent-child dyad the "first and primary intergenerational relationship" (p. 1), Ames (1999) advocates for studying intergenerational relationships in families using a life course approach. She highlights the need to know more about parent-child relationships in midlife and older adulthood because, as society ages, these relationships span more years and are more complex. Scholars know little about these relationships. Generational relations have also been

studied using a life course perspective over time by comparing two cohorts of adults (Hareven, 1994). Taylor, Chatters, and Jackson (1993) documented the strong intergenerational ties of Black families in their Three-Generation Family Study, which derived demographic and relational data from the National Survey of Black Americans, conducted by the Program for Research on Black Americans at the University of Michigan's Institute for Social Research. Greenfield and Marks (2005) used secondary data from the 1995 National Survey of Midlife in the United States (MIDUS) to investigate the relationship between adult children's problems and their parents' well-being. As with other studies of intergenerational relationships, they found that parents' and children's well-being mutually influenced their developmental trajectories. Finally, Umberson, Pudrovska, and Reczek (2010) conducted a review of scholarship on parenthood, childlessness, and well-being also using the life course principle of *linked lives*. They found that parents and adult children maintain relationships and continue to influence, stress, and support one another throughout the life course. In addition to influencing each other's development, intergenerational relationships are key to the transmission of cultural and familial norms from one generation to the next.

Life Course Theory, Linked Lives, and Intergenerational Transmission

Life course theory also has been used to study family change over time, especially concerning the processes and transmission of socially constructed meanings in culturally specific contexts and using the family as the unit of analysis (Bengtson & Allen, 1993). Bengtson and Allen noted, "The family as a social group provides meaning to events; through the perceptions of its members it defines as a reality, enduring and continuing over time, passed on through the lineage (1993, p. 479)." The concept of intergenerational transmission within families is well documented in the literature.

Intergenerational Transmission of Pathology. Though scholars have studied the intergenerational transmission of many concepts, the literature on intergenerational transmission is still saturated with studies of pathology, including incarceration, violence, and divorce.

Incarceration. In a study of the extent to which incarceration "runs" in families,

Andersen (2018) found that women are just as likely as men to contribute to the transmission of incarceration across generations because of their partner choices. Further, children with one or more incarcerated parent are more likely to have at least one other incarcerated family (Wildeman & Wakefield, 2014). Further, estimates indicate the majority of Black men who did not finish high school (70%) are expected to be incarcerated by age 30 (Western & Wildemen, 2009). These findings provide evidence of how social disadvantage proliferates in families, especially among poor, Black families.

Domestic Violence. Ehrensaft et al. (2003) studied the intergenerational transmission of domestic violence with a 20-year longitudinal study of children and found that the strongest predictors of future intimate partner violence were conduct disorder, exposure to domestic violence between parents, and power assertive punishment (e.g., corporal punishment, spanking). However, a child need not develop a conduct disorder to increase the likelihood of domestic abuse (as an abuser or abusee). Exposure to domestic violence between parents was the greatest risk factor to being abused by a future partner (Ehrensaft et al., 2003).

Divorce. Data from the national, longitudinal Study of Marriage over the Life Course indicates intergenerational transmission of divorce regardless of racial group. Black offspring of divorced parents were only slightly less likely to divorce than White offspring (Amato, 1996). Divorce transmission was unrelated to offspring being more accepting of divorce due to personal experience with parents. Contributors to offspring divorce included both offspring spouses

having experienced parental divorce, problematic interpersonal skills (i.e., anger, jealousy, contempt, poor communication, infidelity), and a lack of skills and attitudes that facilitate successful marital relationships (i.e., trust, fidelity, communication). Problematic behaviors associated with divorce were transmitted to offspring, which were associated with their own divorces. The pathological concepts transmitted in these examples do not comprise the whole of scholarly tests of Elder's principle of *linked lives* and intergenerational transmission. Positive attributes also have been found to be passed down through families.

Intergenerational Transmission of Values. Life course theory's principle of *linked lives* also has been used to study the intergenerational transmission of positive behaviors, values, traditions, and parenting practices.

Values and Attitudes. The intergenerational transmission of values was confirmed by a study examining the transfer of mothers' work values (e.g., money, security, autonomy) to their sons and daughters, confirming that mothers and their offspring held similar extrinsic and intrinsic values about work (Ryu & Mortimer, 1996). Adolescents also were more likely to endorse the same attitudes as their parents across six domains: authoritative parenting, conventional life goals, gender egalitarianism, deviancy, abortion, and sexual permissiveness (Schofield and Abraham, 2017). Intergenerational transmission of literacy was confirmed between parent-child dyads (Bus, van IJzendoorn, & Pellegrini, 1995), indicating that shared reading experiences (parent-preschooler reading together) led children's language growth, emergent literacy, and reading achievement. The linked lives principle suggests that parents' values and behaviors influence children's values and shared experiences can impact behavior.

Parenting Behaviors. The parenting one experiences as a child at multiple developmental periods (i.e., in early childhood, middle childhood, and early adolescence) predicts how one will

parent their own three-year old children (Belsky, Jafee, Sligo, Woodward, & Silva, 2005). For mothers, there also is a relationship between the type of parenting experienced and the type of parenting enacted, especially for supportive parenting relationships; there is no effect for fathers (Belsky et al., 2005). Parenting behaviors also are transmitted from parents to children and may be specific to their experiences in adolescence (Chen & Kaplan, 2001). Development is a result of the intersection of individual experiences and their timing in the life span as well as the transmission of values across generations.

Gender Role Attitudes. The timing of the intergenerational transmission of values is sensitive to children's developmental stage (Min, Silverstein, & Lendon, 2012). Using secondary data from the 1971 and 2000 waves of the Longitudinal Study of Generations, Min, Silverstein, and Lendon (2012) found stronger rates of transmission earlier than later in the family life course which then remained stable in children over time. At baseline and follow-up, children held more egalitarian gender role attitudes than their parents.

Gender Role Attitudes Through Parenting Behaviors. Values are transmitted differently across generations and there are differences in active (attitudinal) and passive (behavioral) transmission (Glass, Bengtson, & Dunham, 1986; Roest, Dubas, & Gerris, 2010; Min, Silverstein, & Lendon, 2012). Societal or community values are easier to transmit to children, likely because of reinforcement of these beliefs from multiple groups. Parents, especially mothers, influence their young adult children's gender role attitudes and preferences for housework (Cunningham, 2001). Parental modeling shapes children's attitudes (Cunningham, 2001) and parenting behaviors have more impact than parental attitudes on adolescents' gender role attitudes (Platt & Polavieja, 2016). There have been differing opinions about whether or not intergenerational transmission is stronger between same-sex parents and children (i.e., mothers to

daughters, fathers to sons). Acock and Bengtson (1978) did not find a same-sex transmission of the development of political and religious attitudes in a study of middle-aged parents and young adult children. However, Platt and Polavieja (2016) found a stronger transmission of gender role attitudes through same-sex parent—child dyads than through opposite-sex pairs. Both studies used survey data.

Historical Time and Place and Intergenerational Transmission

Scholars also have used the life course principle of historical time and place to explain the adoption of gender egalitarianism (Min, Silverstein, & Lendon 2012; Cotter, Hermsen, & Vanneman, 2011; Moen, Erickson, & Dempster-McClain, 1997). During the 1970s and 1980s, when gender equality social movements were prominent, society experienced turbulent changes, including the recognition of women's increasing participation in higher education and the paid labor market. Social-historical context sets the frame for the expectations of women and men. Currently, both men and women are expected to be both career and child-focused (Daly, 1996; Daly, 2001). Life course principles have been used to study mothers' feelings of adequacy. A cohort of mostly White middle-aged women who became parents in the 1970s reported using their adult daughters' work and family successes or failures to measure the adequacy of their own lives (Carr, 2004). There have been slightly different expectations of Black families.

Black couples historically have been more likely to be dual-income couples (Raley, Mattingly, & Bianchi, 2006). Some of the reasons for this are that Black women have been "going to work" for longer than White women and are more likely to have been raised by a working Black woman (Beckett and Smith, 1981; Dietrich, 1975; Raley, Mattingly, & Bianchi, 2006; Scanzoni, 1975; Willie & Greenblatt, 1978). Thus, Black mothers have modeled for their sons and daughters that women work outside the home or for wages. In fact, even in the post-

emancipation United States, Black women and men began establishing systems of dual, balanced work (Dill, 1988). Black families are more likely than Whites to feature a woman as an equal provider of income to the family (Raley, Mattingly, & Bianchi, 2006). To that end, dual-income families are not new to the Black community. Sadly, they have not been studied in terms of the negotiation of egalitarian processes to balance work and family.

Historical Context of Black Families

Over the last century, many scholars of sociology and history have chronicled the life experiences of Black Americans, particularly E. Franklin Frazier, Herbert Gutman, and Daniel Moynihan. The history of Black families in the United States primarily has been told by male, and mostly White sociologists. Frazier, the only Black sociologist of this group, successfully covered the first 150 years or so of Black history—the slave trade through the early 20th century—through personal narratives. Frazier's (1939; 1948) efforts to state the facts of Black families are filtered through the zeitgeist of the 1930s, which was particularly critical of the movement of Black families from the rural south to northern cities. Much of Frazier's work labels black families as socially disorganized. Moynihan primarily wrote of the deficits of Black families, pointing to bleak economic prospects for Blacks because of a lack of education, high rates of unemployment, neighborhood poverty, illegitimate children, and primarily femaleheaded households (Moynihan, 1965). Due to his perception of Black men, Moynihan wrote that they could not and, therefore, would not participate in the matriarchal structure of Black families. Gutman's historical tome (1976) reshaped the deficits highlighted by Moynihan and refocused the scholarly conversation about Black families on their abilities to succeed despite environmental and societal challenges, such as enslavement, family disruption, poverty, and migration.

Primarily researched and reported by these male sociologists, the history of the Black family in the United States has included stories from slavery (1619–1863), the experiences of Black families during Reconstruction (1864–1877) and Post-Reconstruction (1878–1915), and tales of Black families during the Great Migration (1915–1970).

Slavery (1619–1863). The Atlantic slave trade began in approximately 1619 with the arrival of the first Africans in Virginia who were cargo on board the *San Juan Bautista*, a ship bound for Mexico, crewed by Spanish sailors. The slave trade was a lucrative business; in 1860, one enslaved Black person could fetch \$800 or about \$130,000 in today's prices (Williamson & Cain, 2016). Enslavers were respected as wealthy, influential businessmen. The slave trade was controversial—condemned by many of those in the cities of the Northern colonies who had no use for slaves but valued by the plantation owners of the Southern colonies who needed slaves to work their fields and farms. An expensive commodity, slaves were closely watched and controlled. The slave trade continued for more than 200 years with approximately four million Africans within the bounds of slavery by the time of the Emancipation Proclamation, which legally ended slavery in 1863 (this point is debatable).

Black Family Life During Slavery. Enslavers, or masters, maintained domination over their property by controlling their most basic human needs. An enslaved person looked to his master for food/water, shelter, clothing, and safety. Enslavers viewed enslaved people as savages or children in need of control who lacked paternal and kinship attachments (Gutman, 1976). On the plantation, many enslavers forbade enslaved people from learning to read and write. Enslaved people were assimilated into the White man's culture through religion and the teaching of White moral values.

In terms of family relationships, there is a difference in what has been reported by White enslavers/legislators and enslaved people themselves. On large plantations, enslaved people were seen and treated much like chattel, purely machines, and relationships between enslaved people were viewed as machine-like and detached from romantic feelings (Frazier, 1939). Moreover, the intra- and interstate slave trade contributed to a lack of marriage or familial ties between enslaved couples as well as between enslaved parents and their children. Civil law did not sanction marriages amongst slaves, and the common myth of licentious black sexuality maintained the need to control the sexual behaviors of Black men and women (Gutman, 1976).

Slaves who worked large plantations were less likely to form strong kinship relationships than were slaves who lived on smaller (more stable) plantations. However, enslaved women played a dominant role in their family and were regarded as having a keen attachment to their biological children while also playing a foster-mother role to the children of her master. The female-headed household dominated Black families during slavery because of the less secure position of Black men. Male slaves were more often traded or sold to other plantations than were female slaves. Enslaved women also had consensual and nonconsensual sexual relationships with their White masters, which sometimes resulted in the conception of mixed race (*mulatto*¹) children. The children of these interracial relationships often were given priority status amongst slaves—demonstrated by working in the house rather than the fields and being taught to read and write. Some of these mixed race offspring assimilated into White culture.

There was a social hierarchy amongst slaves within and between plantations. Lighterskinned slaves worked as personal servants and performed other tasks within the master's home

¹ As a biracial Black woman, I grew up with an extremely derogatory understanding of the word *mulatto*. I understand, however, that this slur is commonly used to describe people of Black and white race ancestry. This is the only place this slur will be used.

and occupied a higher social plane than darker-skinned slaves, who worked in the fields. In addition, society held slaves belonging to wealthier White families in higher regard than the slaves belonging to poorer White families. Moreover, a hierarchy existed between free and enslaved Blacks, with free Blacks having more social status than slaves. Despite this slight difference in social status between free Blacks and slaves, the United States government did not recognize any difference in the rights of Blacks; they simply had none. Dred Scott v. Sandford (1857) exemplifies the perceived status of free Blacks and enslaved Blacks. Dred Scott (an enslaved Black man) and his wife, Harriet (a free Black woman), lived in a free state during their marriage before returning to a slave state. In 1857, just before the Civil War broke out, Dred Scott sued for his freedom, but the Supreme Court decided that Blacks had no rights in federal court and that slave states no longer had to honor the 'once free, always free' clause (Dred Scott v. Sandford, 1857). The Dred Scott decision intensified strains about slavery between the northern and southern states.

Reconstruction (1864–1877) and Post-Reconstruction (1878–1915). Northerners were complicit in the continued domination over Blacks in the years following the end of the Civil War and the end of slavery. There was a common understanding that the South was more equipped to handle the issues with free Blacks (McKee, 1993), which may have been due to southerners' previous experience with controlling Blacks during slavery. In addition, theories that emerged during this time, such as evolutionary theory and Social Darwinism, conjectured that Blacks were less intelligent and inferior to Whites. Andrew Stephens stated, in his 1861 'Cornerstone' speech,

Our new government is founded upon exactly the opposite idea; its foundations are laid, its corner- stone rests, upon the great truth that the negro is not equal to the white man;

that slavery subordination to the superior race is his natural and normal condition. This, our new government, is the first, in the history of the world, based upon this great physical, philosophical, and moral truth...With us, all of the white race, however high or low, rich or poor, are equal in the eye of the law. Not so with the negro. Subordination is his place. He, by nature, or by the curse against Canaan, is fitted for that condition which he occupies in our system.

These pseudo-scientific theories added further support to the belief that Blacks needed supervision and control; theories to which both northerners and southerners subscribed.

The Thirteenth Amendment, ratified in 1866, is primarily regarded as the Constitutional Amendment that abolishes slavery in the United States; however, this amendment further endorses institutionalized control. The Amendment states: "Neither slavery nor involuntary servitude, except as a punishment for crime whereof the party shall have been duly convicted (emphasis added), shall exist within the United States, or any place subject to their jurisdiction" (U.S. Const. Amend. XIII). The fear of Blacks and the need to control them likely led to the period of mass incarceration of Black men in the years after Reconstruction and the further disruption of Black families. The Fourteenth Amendment (ratified in 1868) granted birthright citizenship and equal civil and legal rights to free Blacks and the four million Black slaves emancipated after the American Civil War, and the Fifteenth Amendment (ratified in 1870) gave Black men the right to vote. However, these amendments did not have any impact on the day-to-day lives of Black people.

Because of the widespread belief that Blacks were inferior to Whites, segregation of the races began and was upheld by multiple prominent court cases. One example of this is Plessy v. Ferguson, which upheld racial segregation laws or "separate but equal" in the United States. The

Court opinion issued by Brown stated, "The assumption that the enforced separation of the two races stamps the colored race with a badge of inferiority. If this be so, it is ... solely because the colored race chooses to put that construction upon it" (Plessy v. Ferguson, 1896). This statement legally absolved White Americans of any responsibility for the beliefs of racial hierarchy that had been purported by evolutionary and Social-Darwinist theories during this period. Instead, the Supreme Court shouldered the colored race with constructing how they might be inferior (Plessy v. Ferguson, 1896). These ideologies were used to problematize social conditions among Blacks, including poverty, female-headed households, birth rates, life expectancy, low educational attainment, and crime (Frazier, 1939). Plessy v. Ferguson further solidified the practice of segregated facilities, including public bathrooms, water fountains, and schools.

In the years following the Civil War, violence and intimidation were used to prevent Blacks from voting, especially in the southern states of South Carolina, Florida, and Louisiana. Despite the formal Constitutional rights documented in the Thirteenth through Fifteenth Amendments, segregationist or "Jim Crow" laws governed life in the south. The legality of "separate but equal" was finally overturned in 1954 with the Supreme Court Case Brown v. Board of Education of Topeka. One could argue that Blacks did not actually begin to enjoy the freedoms afforded to them by the Thirteenth, Fourteenth, and Fifteenth Amendments until after the civil rights movement or approximately 1960, nearly 100 years after the Emancipation Proclamation was issued in 1863.

Black Family Life During Reconstruction and Post Reconstruction. During this post-Reconstruction period, Blacks were no longer enslaved by slaveholders, but they continued to be confined by sharecropping and debt. Sharecropping required that Black families rent small

plots of land, which they worked themselves. At the end of each year, sharecroppers would pay the landowners a portion of their crop. The inability of Blacks to own their living space was perpetuated through World War II with the failure to issue mortgages to Black soldiers returning from war and through the predatory lending practices of more recent years. Overall, Black families recently freed from slavery experienced turmoil during the post-emancipation period.

Black families who moved off the plantation were displaced and had to figure out how to survive and reconstruct their families. Throughout the history of Black families in the United States, Gutman (1976) argues that members of the Black family were constantly connected despite slavery and the Reconstruction period and separated family members were committed to reconnecting with one another. The stories recounted by slaves to Frazier (1939), including the practice of tying the hands of children together and walking for miles when leaving their plantations and using personal networks to track down family members who were sold off the plantation, demonstrate former slaves' commitment to their families.

Great Migration (1915–1970). Historians date the Great Migration to about 1915 through 1970 but for some, like my great-grandfather, the move was earlier. During this time, Black families residing in southern states moved in droves to northern and western cities, such as Chicago, New York, Washington D.C., Los Angeles, Detroit, Oakland, and Philadelphia (Wilkerson, 2010). Often referred to as migrants, a better description of these Black settlers may be asylum seekers, those who are looking for protection and security from their country (or state) of origin. In the case of Black families migrating north and west, these families were looking for freedom and opportunity, higher wages to afford them the ability to provide for their families, and the security of life without the fear of lynching. Wilkerson defines this collective, leaderless movement of Black people as the "first mass act of independence by a people who were in

bondage in this country for far longer than they have been free" (2010, p. 10). In all, more than two million Black people moved north and west during the Great Migration with hopes of opportunity (McKee, 1993).

Life in the northern cities was only slightly less troubling than life in the south. About forty years after emancipation, Du Bois (1903) wrote, "...the freedman has not yet found in freedom his promised land" (p. 6) and identified that the color line was the problem of the twentieth century; he was correct. A color line existed in the neighborhoods of the cities experiencing an influx of Black families (McKee, 1993; Wilkerson, 2010). Organizations in Harlem such as the Save-Harlem Committee and the Harlem Property Owners Improvement Corporation sprung up to address the issues of a growing colored population in New York. Real estate agents were advised not to rent to Blacks in particular neighborhoods. The segregationist movement was afoot in Northern cities; however, segregation did not take hold because of the economic demand for housing and the need for revenue (Wilkerson, 2010). White families were leaving Harlem because of the influx of Black families, and White landlords could not afford to lose tenants. Capitalist principals won out over prejudice, and White landlords opened their apartments to Black families and charged higher prices (McKee, 1993).

Black Family Life During the Great Migration. After the end of slavery, Black people established their families within the expectations of society. The role of the man in the home may also be attributed to the shade of his wife's skin, with mixed race Black women often the "object of pleasure and display" (Frazer, 1939, p. 439) signifying economic success. Colorism, or the prejudices and beliefs that lighter-skinned Black people were more economically and socially desirable than darker skinned Black people, had much to do with the Black family's economic and social advantages (Burton, Bonilla-Silva, Ray, Buckelew & Freeman, 2010; Few, 2007).

Colorism beliefs were held over from the skin color hierarchy introduced in slave times.

Colorism also was associated with Black males' role in their households, with status given to Black men with lighter-skinned wives (Frazer, 1939). Further, Frazier argues that Black men's assuming more authority in the family is correlated with male interest in the family and family sustainability. Male interest in the household may have increased when he was married to a light-skinned Black woman whose presence in the family suggested his economic success.

After the Civil War, Black men temporarily reassumed their role as the head of household (Frazier, 1939), as opposed to the matriarchical family structure under slavery. The disorganization of Black families defined by Frazier stems from economic instability experienced in the Great Migration to big, urban cities; this disorganization resulted in Black women reassuming the dominant role in the family. Frazier states that the Black wife "continues to occupy a position of authority and is not completely subordinate to masculine authority even in those families where the man is present" (p. 461). In 1965, Daniel Moynihan further positioned the Black woman as the head of the Black family in a controversial report he authored for the United States Department of Labor (Moynihan, 1965). This positioning branded Black families as ill-suited for the patriarchal U.S. societal norms because of the matriarchal structure of Black families. This matriarchal structure can trace its roots to slavery.

Egalitarian Families

Another key concept in the conceptual framework of this study is egalitarianism. As mentioned above, Black couples historically have been more likely to be dual-income couples (Raley, Mattingly, & Bianchi, 2006) and have negotiated balanced (Dill, 1988) support of the family. Egalitarianism is socially-constructed and based on individuals' perception, comprehension, and interpretation of the world. In the context of this manuscript, egalitarian

refers to shared humanity, ability, and responsibility across gender lines, within marriage and/or between co-parents. Masterson and Hoobler (2015) developed family identity-based typologies of dual-earner couples (e.g., *traditional*, *non-traditional*, *family-first*, *outsourced*, and *egalitarian*), describing egalitarian couples as follows:

...Egalitarian couples will strongly incorporate both care- and career-based roles into their family identities. Couples structured in this way enable care and career in both partners' family identities simultaneously...In considering the shared and strong emphasis on care- and career-based roles, both partners will engage in a range of activities inside and outside the home to help verify the ways in which they construe their family identities...Egalitarian couple structures reflect societal shifts in gender equality—as family identities defined in terms of care and career are perceived to be gender neutral. At the same time, in considering the strong incorporation of both care and career into both partners' family identities, there is likely less clarity regarding role responsibilities (as compared with "traditional" and "non-traditional" couple types; p. 83).

In an egalitarian family, partners identify a balance between work and family responsibilities and maintain a satisfactory relationship between them (Thornton and Young-DeMarco 2001).

Families who take an egalitarian approach to parenting attempt to equally share the tasks required to take care of the household and children as well as earn income through employment.

The concept of egalitarian parenting has been raised in recent research (Perry, Harmon, & Leeper, 2012; Thornton & Young-DeMarco, 2001; White, 2006a; White, 2006b) but has not focused exclusively on dual-income Black families. Related to the issue of dual-income families is the concept of the egalitarian family, which counters Western traditional patriarchal family structures where the father is the breadwinner and the mother is the caretaker. More Black men

than White men have egalitarian views toward gender roles particularly related to child rearing (Glauber & Gozjolko, 2011; Kane, 2000; Orbuch & Eyster, 1997). Moreover, recent research indicates that Black fathers are more involved in their children's daily care than are White and Hispanic fathers (Jones & Mosher, 2013).

Egalitarian parenting has been called different things overtime. Research from previous decades referred to this concept as 'equalitarian' finding that patterns of power relationships among Black married couples were primarily viewed as partnerships with shared authority and neither spouse being more dominant than the other (Mack, 1978; Middleton & Putney, 1960; TenHouten, 1970; Willie, 1976). The cultural schema of work–family integration (Dean, Marsh & Landry, 2013) also is similar to egalitarianism. Work–family integration refers to a mother's belief that her career is equally as important as her husband's and perceives herself as both worker and mother (Dean, Marsh & Landry, 2013). In addition, egalitarian parenting has been described as 'feminist parenting' because it challenges societal norms that dictate that men and women must adhere to traditional gender roles (White, 2006a, 2006b).

Egalitarian attitudes about parenting and household management may occur within married or cohabitating couples; enacting an egalitarian family does not require the sanctimony of marriage. Egalitarian parenting may be related to cohabitation statistics in that those with a more "progressive" view of coupling (i.e., living with a partner without the sanctimony of marriage) may be congruent with a more "progressive" view of rules and roles, specifically gender roles (Perry et al., 2012). Progressive gender roles may be indicative of having more flexible definitions of parenting responsibilities, which could facilitate the potential for more egalitarian approaches to parenting. Similarly, Black family patterns are distinctive and suggest greater acceptance of alternative family arrangements (Carlson, VanOrman, & Pilkauskas, 2013).

In addition, young couples also are more willing to create egalitarian relationships and an egalitarian family (Thornton & Young-DeMarco, 2001). These studies suggest that young Black couples may be more likely than others to engage in egalitarian families.

Family processes in the 1970s-1980s. Multiple studies examined gender role attitudes in the 1970s and 1980s. A comparative study of females' attitudes regarding women's participation in the labor force found that daughters were more egalitarian in 1980 than their mothers were in 1962 (Thornton, Alwin, & Camburn, 1983). Black families were more likely than White families to exhibit egalitarian roles, sharing responsibilities for earning wages, maintaining the household, and caring for children (Beckett & Smith, 1981). Black parents were also more likely than White parents to enact egalitarian practices, regardless of income level (Dietrich, 1975; Scanzoni, 1975; Willie & Greenblatt, 1978). Black women more often than White women rejected the dominant culture's views of gender roles, which has been attributed to Black women's long history in the workforce: "within Black culture, wage labor has been a normative and integral component of womanhood" (Dugger, 1988, p. 439).

Differences in Black and White men's attitudes about gender roles over time and relative to their mothers' gender role attitudes have been documented using linked longitudinal motherson data from 1967 to 1981 (Blee & Tickamyer, 1995). Black men were more likely than White men to have more liberal attitudes toward women in the workforce and were more likely to have personal experience with a working woman (Blee & Tickamyer, 1995).

The changing role of men during the 1970s and early 1980s was concurrent with the construction of "new fatherhood" (Messner, 1993). A 'new father' prioritized family, exhibited by an emphasis on childcare, over career goals. This period also saw the introduction of multiple constructions of masculinity that allowed for "softer" (Messner, 1993, p. 725) or egalitarian

representations. Further, the construction of "new father" and "new man" was based on the experiences of middle and upper-class, White heterosexual men (Messner, 1993). There is little information about Black fathers during this time in American history. More recently, "new fathers" have been described as men who maintain egalitarian attitudes toward the family roles of men and women (McGill, 2014). These men demonstrate nontraditional fathering attitudes, work more than 40 hours per week, and contribute to the care of their children in a manner that is distinctive from those with traditional fathering attitudes who worked similar hours (McGill, 2014). There is a subset of "new fathers" who exhibit nontraditional attitudes about fatherhood and spend more hours per week, on average, with their children than do fathers with traditional attitudes (McGill, 2014). However, the men in McGill's study were mostly White. "New fatherhood" is not new for Black fathers; however, Black fathers have not been exclusively studied.

Dual Earner Families

Dual-earner couples have been studied to understand how they define, discuss and negotiate time, and maintain family schedules and responsibilities (Becker & Moen, 1999; Curenton, Crowley, & Mouzon, 2018; Daly, 2002; Dean, Marsh, & Landry, 2013; Ray & Jackson, 2013). Couples use informal and formal negotiation processes that emphasize communication, cooperation, and tradeoffs (Daly, 2002). Couples also seek to achieve fairness and balance, or egalitarianism, in the distribution of tasks. Couples report using "trading off" or "tag team" strategies to alternately scale back or intensify their focus on career (Becker & Moen, 1999). "Trading off" strategies promote egalitarianism in theory; however, in practice, gender implications are uneven, with more women than men scaling back their professional goals and opportunities (Becker & Moen, 1999). Other couples highlighted the need for communication,

negotiation, juggling, and sacrifice to manage their work-family lives (Ray & Jackson, 2013). The ability to manage and negotiate shared responsibilities requires a strategic construction of couples' partnerships (Becker & Moen, 1999). These studies also only included White couples.

How couples negotiate time is also related to cultural norms. In negotiations, couples often refer to prior experiences, which may have been shaped by gender roles and societal expectations (Katovich & Couch, 1992). These negotiations must be attentive to what has been done previously and reconciled with what needs to be done going forward. Employment statistics from 1960 to 2005 show that Black mothers consistently have had higher rates of employment than White and Hispanic mothers (Cohany & Sok, 2007; Guy & Fenley, 2014; McKay, 2007). It may be assumed that, since Black women have been active in the labor force since 1890, Black couples have more experience negotiating time and managing family schedules. Black mothers have described how they share parenting responsibilities with their spouse by coordinating schedules and dividing duties (Curenton, Crowley, and Mouzon, 2018). However, only a few studies have examined the processes by which Black families negotiate partnerships, share household and childcare responsibilities, and manage time (Curenton, Crowley, & Mouzon, 2018; Dean, Marsh, & Landry, 2013).

Where Do I Fit In This Narrative?

I came to the topic of egalitarianism in Black families through personal experience. I grew up in a northeast college town with both parents, a brother, a sister, a cat, and a dog. Both of my parents worked and earned a middle-class income. My mom, a White waitress who attended college in the evenings, was studying to be a nurse. My dad, a Black former Army Lieutenant who worked in sales, was our primary caregiver when Mom was in class or studying. In the evenings, Dad made dinner and did the dishes. He checked homework, coached Little

League, and was the only member in the audience for our choreographed dance routines (most often to Paula Abdul's greatest hits). My perception is that my childhood experience was not unlike that of my cousins and friends. I thought we were a pretty normal Black family...but I may have been wrong.

Twenty-five years later in graduate school, I chose an article to critique for a core course in my Family Science and Human Development doctoral program. The article (McGill, 2014) examined "new fatherhood"; a newfangled idea that men (predominately White men) provided care for and spent time with their children in addition to working full-time hours. I was blown away. I could not believe "new fatherhood" was new. It certainly was not new to me and I did not think it would have been new to my brother, cousins, uncles, friends, and husband – all of whom had experienced or had been enacting "new fatherhood" for an extended period of time. History tells us that Black families have operated as dual-earner households since the post-Civil War Reconstruction era (Frazier, 1948; Moynihan, 1965). As such, Black couples have had to share the responsibilities for financially and emotionally providing for their families. I could not understand why the fathering practices of Black fathers had not been highlighted in this new conceptualization of fatherhood. I felt that a piece of the story was missing.

Since I began reviewing research studies on Black families, I have been disappointed to find that many are written from a cultural deficit perspective and focus on the failures or pathologies of Black parents. Fatherhood scholars have written about absent, nonresident, unmarried, and incarcerated Black parents (as discussed in: Frazier, 1948; Moynihan, 1965). These narratives do not reflect my current or past experiences. I have been looking for strengths-based research that focuses on the assets of Black families and resiliency of Black fathers who take care of their families despite the institutional odds, including racism and emasculation. I am

using my dissertation to tell the strengths-based story myself, through the perspectives and memories of four generations of my family.

I aim to study the construction and transmission of egalitarianism across four generations of my family—from my grandparents (Generation 1 or G1) to my parents (Generation 2 or G2) to myself (Generation 3 or G3) to my children (Generation 4 or G4). The primary research question this study seeks to answer is: How did the social-historical context, as filtered through previous generations, influence my conception and practice of egalitarian parenting? Further, because much of the research on the intergenerational transmission of gender role attitudes is based on secondary quantitative panel data, this qualitative study collected the narratives of four generations of my family, using the myself as the unit of analysis. This dissertation is a story of how my family constructed and transmitted egalitarianism, which in turn influenced my individual development.

CHAPTER 3: METHODS

Introduction

The purpose of this study is to explore how my Black family constructed and transmitted egalitarianism across generations, with special attention to the social movements of the 1960s. Life course theory and its principles of *linked lives* and *historical time and place* frame this study. This autoethnography included critical reflection, journaling, and in-depth individual oral history interviews with 13 family members across four generations. Trustworthiness was ensured by using member checks and consultation with critical friends. This study concludes with the analysis of the experiences of four generations of my Black family, allowing the multigenerational voices to describe their experiences.

This study focuses on the construct of egalitarianism in the lives of an intergenerational network of great-grandparents (G1), grandparents (G2), parents (G3), and children (G4) in a Black family. Moreover, I initially suspected that the current and prospective parenting practices likely passed through generations of this family have been influenced by the social-historical context of the 1960s and 1970s. I expected that the social-historical milieu of this time in history would explain how context impacted G1's development into adulthood and shaped their parenting practices, which were then transmitted to and between their descendants. Further, I anticipated that, within my family, egalitarianism was constructed, modeled and passed down from G1 to G2 to G3. Because of this, G4 has experienced egalitarian parenting (from G3) and expects to enact similar practices in their future familial relationships. In this autoethnography, I represent G3. For G3, I integrate my perspective with that of my husband, siblings, and a cousin.

The organization of research participants into age-graded groups aligns with Elder's life course perspective as individuals' lives are synchronized with their generations' experiences in

relation to historical events. These three clocks (i.e., ontogenetic, generational, and historical) are interrelated and mutually influencing. I believe that G1, raised during the Great Depression and World Wars I and II, and G2, raised in the 1950s and 1960s, likely experienced more gender-specific work and family roles than did future generations. G3, raised in the 1970s and 1980s, likely experienced less gender-specific work and family roles and benefited from greater educational and occupational opportunities as a result of the civil rights and women's movement of the 1960s. G4, raised in the 2010s, experienced very little gender-specific work and family roles and might expect the same for their future families. Finally, I suspected that each generation experienced egalitarianism to an increasing degree.

Hareven (1994) makes clear the distinction between 'generation' and 'cohort'. The term *generation* is used to describe genetic or fictive kin (non-blood relative) relationships within a family (i.e., grandparents and grandchildren, parents and children). *Cohort* refers to the agegraded group of individuals born in a span of time that has shared particular social-historical experiences. Cohorts are nonhomogeneous as they are each influenced by historical events and cultural traditions particular to their age-graded group (Hareven, 1994). A generation might include multiple cohorts and the two terms should not be confounded (Elder, 1978; Hareven, 1994; Kertzer, 1983). This study uses the term "generation" to describe groups of individuals within my family who were/are parents at about the same time. This study's sampling frame includes generations of genetic or fictive kin bound within particular age ranges, so that they share particular social-historical experiences.

Hareven (1994) indicates that use of social-historical perspective allows researchers to do three things: 1) compare past and contemporary phenomena, 2) examine change over time, and 3) identify models of coping that may be adapted for use in the present. This study used

narrative inquiry to accomplish all three. I collected the retrospective life history of G1 and G2, the current experiences of G3 and G4, and the prospective experiences of G4. Elder (1998) recommends the use of retrospective life history methods for "recovering knowledge about the enduring effects of past events" (p. 2). In their study of the Chinese Cultural Revolution, Elder, Wu, and Yuan (1993) surveyed young people involved in the crisis of Tiananmen Square about their lives prior to that historic event.

The American social movements of the 1960s, including the Civil Rights Movement and Women's Liberation, comprised the social-historical context of G2 parents. This study examined the social-historical context of G1 as parents. Considerations of the social-historical milieu helped to explain how context impacted G2's development into adulthood and shaped their parenting practices. In particular, it was important to understand how this context shaped egalitarianism in Black families.

In my family, I studied my paternal lineage. Figure 1 shows a model of the four generations that were included in this study. In Figure 1, the members of my family who are deceased are marked with a (D), including my father and his parents. In order to include their voices in this study, I relied on the input of those who knew them in life (e.g., spouses, siblings, children) and my own memory. I gathered the perceptions of multiple individuals in my family about the construction of egalitarianism through oral history interviews. The decision to provide the voices of the deceased with those who knew them in life is supported by African concepts of time and space like *sasa* and *zamani*. As previously argued, life course theory is guided by temporal and positional concepts. Similarly, African religious and philosophical concepts of time and space are guided by the concepts of sasa and zamani (Mbiti, 1969/1997).

Familial relationships, knowledge of ancestors, and oral history remain important to Black families in the United States. This autoethnography included oral history interviews with individuals, still living in the *sasa*, who knew the deceased during their lives. The oral history interviews collected stories and narratives from participants. During the interview, a photograph was shared with participants of my grandmother at the White House in 1965.



Figure 1. Priscilla Sisco Swann standing in front of the White House

Research Design and Methodology

I used autoethnography to tell the story of my family. The research question was addressed using multiple qualitative data collection methods, specifically in-depth individual oral history interviews and personal critical reflections recorded in audio journals. Qualitative research can facilitate the understanding of family processes, such as the construction, practice, and transmission of egalitarianism (Daly, 2007). These methods allowed me to tell the story of egalitarianism in my family using multiple voices, including my own. In this chapter, I use the term *oral history* to define what I collected from others to supplement my memories and delve deeper into stories that I may have heard only as a child. As a researcher living out the

egalitarianism I learned from my parents and grandparents, I realize I have a new perspective on what was modeled and passed down to me and what I am modeling and passing down to my children. In this study, I occupy more than one space; I am simultaneously the researcher collecting the stories and memories of others and a subject of the research, using reflexivity to provide an insider view of the transmission process as I experienced it.

My individual development is the focus of this autoethnography. This study examined how my family constructed and transmitted egalitarianism, which in turn influenced my individual development. I collected data from the four generations I have known in the *Sasa*. This study required considerable vulnerability; however, Philaretou and Allen (2006) indicate that this method can be liberating, especially for researchers from marginalized groups. Though the methodology collected multiple perspectives, the composite of these voices provided a rich account of the culture of egalitarianism in my multigenerational family.

Autoethnography

Autoethnography is a research method that provides space for a researcher to study her own cultural experiences. It emphasizes personal, subjective reflections on experience rather than objective, neutral accounts. Adams and Manning write, "...Autoethnography can allow researchers to offer insider accounts of families; study everyday, unexpected experiences of families, especially as they face unique or difficult situations; write against limited extant research about families; and make research more accessible to nonacademic audiences" (2015, p.1). Autoethnography combines cultural analysis and interpretation (whether culture is "out there" or "in here") with narrative details (Chang, 2008). Autoethnographers analyze their own autobiographical data with the same critical interpretation as they do data from others. The goal

of an autoethnography is to have a cultural understanding of the phenomena based on the perspectives of self and others.

Autoethnography began as a qualitative approach that used anthropological ethnography to study a group of which the researcher is a part, or "one's 'own people'" (Hayano, 1979, p. 99). More recently, autoethnography has moved to a combination of ethnography and autobiography (Humpreys, Coupland, & Learmouth, 2015). Autoethnography is key to critical resistance work as it gives subjugated and marginalized groups the space to tell their stories (Pratt, 1992) and provides an opportunity for multiple voices and perspectives to be given legitimacy (Ellis & Bochner, 2000).

This autoethnography documented my family's construction and transmission of egalitarianism combining personal experience, memory, and storytelling with existing research, theory and cultural analysis of what it means to be a Black family living through the Great Depression, Civil Rights and Women's Liberation movements, Mass Incarceration, and Black Lives Matter.

In addition to answering the question, "Who am I to this study?", this study forced me to both reflect on and report my personal experiences as well as the stories shared by my family. Though I am embedded in my family's culture of egalitarianism, it was necessary to carve out a space to study this culture. Autoethnography requires that the researcher critically reflect on her own positionality and social location. Autoethnography has restorative aspects and can serve as form of healing (Mizzi, 2010). This study was cathartic for me and my family and helped all of us to better understand the deceased family members who were key to the transmission of egalitarianism.

Oral History

To effectively translate the lived experiences of my family, I conducted interviews with family members across four generations. I refer to these interviews as a collection of oral history. "Memory is the core of oral history, from which meaning can be extracted and preserved" (Richie, 2003, p. 19). Oral history involves interviewing people to gain an understanding of their lived experiences. The Black community uses oral history to pass down stories from generation to generation. Oral history also has been used successfully to capture the stories of Black women (Armitage & Mercier, 2009; Brown, 1997; Tuner, 1997; Vaz, 1997; Williams, 2004). In-depth individual oral history interviews allowed me to capture my family's stories of egalitarianism, describe the reciprocal multigenerational relationships, as well as document the interplay between individuals' experiences and historical time and place (Clandinin & Connelly, 2000; Connelly & Clandinin, 1990). Congruent with African religious and philosophical concepts of sasa and zamani (Mbiti, 1969/1997), I collected the oral histories of family members who knew my father and grandparents during their lives to provide my father/grandparents' experiences with egalitarianism and the modeling and transmission of this value from one generation to the next.

Procedure

Participants

I conducted oral history interviews with 12 family members and reviewed five oral history interviews that had been conducted by historians most of whom knew my father and grandparents in the *Sasa*, for the New City Library Oral History Collection. One of my sons never met my father and neither of my sons knew my grandmother. Table 1 displays the people

who were recruited for oral history interviews and the deceased individuals whose perspective their stories may represent. My voice represents G3.

Table 1: Participants

Participant	Generation (G)	Relationship to G3
1	G1	Great Aunt
2	G1	Great Aunt
3	G2	Mother
4	G2	Uncle
5	G2	Uncle
6	G2	Uncle
7	G2	Uncle
82	G2	Cousin
9	G2	Cousin
10	G3	Self
11	G3	Spouse
12	G3	Sibling
13	G3	Sibling
143	G3	Cousin
15	G4	Son
16	G4	Son

² This family member had a close relationship with and operated a small business with G1. ³ This family member was raised by and provided end-of-life care to G1.

I obtained approval from the Montclair State University Institutional Review Board (IRB-FY19-20-1590). All IRB-approved consent forms and interview protocols, and recruitment text can be found in Appendices A-C.

Recruitment

A total of 14 family members were invited to an interview. All participants were contacted by phone, by email, or in person to recruit them to participate in the study. Most of the interviews were conducted in-person at the participant's home between December 2019 and January 2020. Three interviews were conducted by phone.

Recruitment Ethical Concerns

Prior to data collection, I was concerned that some participants would feel undue influence to participate in my study because I am a member of their family. I attempted to mitigate feelings of undue influence by collecting informed consent virtually rather than inperson. The consent form was available online, using SurveyMonkey, and was completed by the participants prior to scheduling the interview. The MSU IRB approved the consent process for all participants, including my own minor children, ages 7 and 11. All recruited participants had the opportunity to review the informed consent and determine whether or not they wished to participate. By removing myself physically from the consent process, I hoped to mitigate any pressure or undue influence the prospective participant might feel. In addition, all recruited participants had time to seek advice from others if desired. I informed all participants that their identities would be protected by the use of pseudonyms and, during the data analysis phase, I used member checks to ensure they approved of my using any data that might be identifying.

Importantly, the oral history interviews I conducted follow the ethical principles and practices of the Oral History Association (OHA, n.d.), which states "Everyone involved in oral

history work, from interviewers and narrators to archivists and researchers, becomes part of a web of mutual responsibility working to ensure that the narrator's perspective, dignity, privacy, and safety are respected." The ethical practices espoused by OHA focus on respect, transparency, awareness of power dynamics, and safety. As an interviewer who collected the histories of my own family, I recognize my power both as the interviewer and as an aspiring researcher. I informed my participants of the purpose of the interview, ensured they were giving their consent voluntarily, and reiterated that they could withdraw from the interview or refuse to answer a question at any time. I respected and honored my family members' participation by allowing them to review the interview (recording and/or transcript) and approve what was included before I analyzed it.

Data Collection

Data collection methods for this study included individual oral history interviews and a researcher journal.

Oral History Interviews

To collect the stories of my families' experiences with egalitarianism, I conducted 12 oral history interviews, reviewed five previously conducted oral history interviews, and recorded my own critical reflections, totaling 775 minutes of shared memories, storytelling, and reflection. Inperson interviews were an average of 54 minutes long. At each interview, participants were asked to confirm their consent to be interviewed and recorded. A semi-structured interview protocol guided each interview, as determined by the generation each participant represented. Interview protocols are provided in Appendix A. Photographs were used to spark conversation and were shared in person or electronically during each interview. Oral history interviewing requires that the interviewer allow the participant to tell their story (Miller & Crabtree, 2004) so

interruptions and requests for clarification were limited. The stories participants told reflected the details they deemed most important to include. After each interview, I documented my reflections and sent the audio recordings to be transcribed by a transcription service (i.e., Rev.com). Some of my family members previously participated in oral history interviews. The Library Association of Rockland County, as part of the Sound and Story Project of the Hudson Valley's Cube, has published five oral history interviews with two of my great-aunts (Peaks, 2014a; Peaks, 2014; Crawford, 2017a; Crawford, 2017b; Crawford, 2017c). These interviews concern women's rights and work as well as perspectives of growing up in the 1930s and 1940s. The archive states that these audio recordings may be used for scholarly or educational purposes. These interviews and transcriptions were included as data in this study.

Researcher Journal

I documented my memories and reflections in a researcher journal that I kept over the course of this study – from recruitment to analyses – as well as thoughts and ideas prior to determining a final topic. Journal entries begin in May, 2017 and end in February 2020. I used an audio recorder or the *Easy Voice Recorder* app to collect these thoughts. These audio files were also transcribed by the transcription service. In total, I recorded 89 minutes of my own reflections, musings, and memories.

It was necessary for me to investigate my suppositions and assumptions by considering my positionality in relation to the data I collected and reviewed (hooks, 2004; Luttrell, 2005, 2010a, 2010b; Sosulski, Buchanan, & Donnell, 2010). After each interview, I determined how my memories of events or stories I had been told might be challenged by what I heard from my participants. Because of my closeness to the participants and the stories they shared, I needed to be open to alternate understandings of events with which I was familiar, including the possibility

that my family may not have been as egalitarian as I perceived it to be. I also was impacted by the cumulative effect of the interviews. I did not intentionally order my participants and allowed convenience to guide the order of interview scheduling. As a result, I was a different instrument between interview #1 and #12. This means that, as an instrument or a tool used to gather data, I had different understandings of the events being described by the first participant and the last. My openness to what I was hearing may have changed between the first and final interview. Moreover, though I collected my own reflections along the way, by the end of data collection I knew more about the events about which I was asking.

Positionality

Prior to conducting these interviews, I disclosed to my participants that I wanted to learn more about their experiences growing up and that I am invested in the stories of Black families. I shared that my identities include granddaughter, daughter, sister, niece, cousin, wife, and mother. I also disclosed that it is important to me that their stories are told, heard, and recorded. For the sake of transparency, I also disclosed that I am interested in the concept of egalitarianism in families. This disclosure may have "biased" my analyses but bias depends on the interest being served (White & Dotson, 2010). I believe that any bias revealing my research questions may have posed is to the benefit of the marginalized group – Black families.

Data Analyses

Gilligan's Listening Guide

I used Gilligan's Listening Guide (LG), sometimes called the Voice-Centered Relational Approach, to analyze the interviews (Brown & Gilligan, 1991; Gilligan, 2015; Gilligan, Spencer, Weinberg, & Bertsch, 2006). LG is designed to facilitate attending to multiple voices within individuals. The multivocal nature of this autoethnography makes LG an appropriate approach

for this study. Adapted from Gilligan (2015), Woodcock (2016) provides a procedural guide that can be used to maximize the goal of LG to focus on the researcher-participant relationship and the interplay between voice and silence, and tune-in to marginalized voices. LG provides a framework that has been used to analyze oral history interviews (Baird, 2012; Golding & Hargreaves, 2018), allowing the researcher to adapt or modify the LG approaches to better focus on participants' voices and meanings. LG requires that each interview be transcribed and listened to (and simultaneously read) three times. Each listening/reading requires a particular focus:

- 1. Listening #1 What are the psychological features of this particular terrain?
 - a. Listening for plot or the stories that participants share
 - b. Listening for silence or pauses or whispering/voice-lowering or trailing off
 - c. Listening for the narrator's response to or being mindful of the interviewer's responses, questions, and confusions
- 2. Listening #2 How does the "I" or first-person voice move across this terrain?
 - a. Listening for "I-statements"
 - b. Listening for the self-voice of the participant or how he or she speaks about himself or herself and in relationship to others
- 3. Listening #3 What voices within the transcript or text speak to or inform the researcher's question?
 - a. Listening for responses to the study's research question
 - b. Listening for the contrapuntal or themes that complement or compete with each other

Because this study focused on the transmission of values and behaviors across generations, the LG was modified slightly for this study to ensure better analysis of the data

across individuals and between generations. The procedures for Listening #1 mostly remained the same as Gilligan (2015; Gilligan et. al., 2006; Woodcock, 2016) prescribed. Interviews were listened to and transcripts simultaneously were read to hear: 1) the stories that participants shared, 2) silence, hesitancy, pauses, or whispering/voice-lowering or trailing off, and 3) the extent to which the participants were mindful of the interviewer's responses, questions, and confusions. In addition, exhibitions of emotion (e.g., crying, deep breaths) and requests for more time before continuing their responses were considered during Listening #1. Similarly, Listening #2 also primarily followed Gilligan's protocol. Interviews were listened to and transcripts simultaneously were read to hear: 1) the "I" or first-person voice of participants' stories and 2) how he or she speaks about himself or herself and in relationship to others. Listening #3 remained unchanged from Gilligan's procedures (2015). Interviews were listened to and transcripts simultaneously were read to hear: 1) stories related to the research question and 2) any statements or themes that are interacting or competing. I made one significant change to the protocol.

Pronoun-Poems

In addition to listening for "I" statements in Listening #2, I listened for other pronoun-based statements (hereafter referred to as pronoun-poems), including he/him (my great-grandfather, grandfather, father, or husband), she/her (my great-grandmother, grandmother, or mother), they (my grandparents or parents), we/us (our family, my uncles, my siblings), and you (Rebecca). These poems are short excerpts from the participants' transcribed narration. For the most part, pronoun-poems were extracted from multiple participants representing a particular generation. Pronoun-poems were co-constructed between the narrator and listener based on oral history concepts, including interpretive authority (Borland, 1998), shared authority (Frisch,

1990) and collaborative authorship (Ochs & Capps, 1996). Portelli states, "There is no oral history before the encounter of two different subjects, one with a story to tell and the other with a history to reconstruct" (1998, p. 28). Based on these oral history concepts, I selected and used the multiple participants' actual language to form poems that would capture the sentiments of one generation. These excerpts include participants' perceptions, memories, values, or beliefs and were curated based on what I understood from them. For example, this poem emerged from statements about a woman in Generation 2:

She's very cerebral.

She looks at everything and takes it all in.

She had to be that way.

She had three kids.

It was not man-to-man coverage,

Like zone coverage.

She needed to be methodical.

She had to do all this stuff,

And still go grocery shopping.

The pronouns of interest described the main protagonists of my study – my family atlarge, my great-grandparents, grandparents, parents, husband, self, and children. In other words,
for each interview, I listened for how the participant described or told stories about others. I
combined these pronoun-statements into a multi-tab spreadsheet. In this spreadsheet, worksheet
tabs were named for each person or couple that was discussed in the interviews. On each
worksheet, I recorded the name of the participant, the timestamp of the statement, and the
quotation. When analyzing qualitative data, researchers should not assume familiarity with their
participants (hooks, 2004); creating pronoun-statements and analyzing them across participants
helped me to focus on the specific people participants described and provided me space to
consider how those descriptions did or did not influence my understanding of that person. At
first, I primarily reduced participants' statements to the pronoun-verb pair in the sentence (e.g., I

went, she wanted, they were), but found that this reduction often did not provide enough context for what the participants' had described. These slight modifications to the LG helped me to focus on participants' stories and perceptions of individuals and our family at-large.

After the pronoun-based statements were extracted from each participant's transcript, they were combined for each of the characters. I created a file of all of the statements about my great-grandparents, grandparents, parents, my husband and myself, and our children, as individuals and couples, respectively. Each family member's perspective was important to understanding how egalitarianism was constructed and transmitted, and which aspects of egalitarianism persisted over time. Creating a repository of pronoun-based statements helped me to develop a greater understanding of my family from the perspective of multiple participants. Pronoun-based statements for each character were analyzed using open-coding. I used an inductive approach to analysis and allowed the data to shape how the findings should be reported. I reviewed the data across respondents to look for a common thread of the construction of egalitarianism, including how this thread was adapted as it moved from generation to generation. The purpose of the analysis was to connect individuals' experiences to greater cultural issues (Wall, 2016). Then, I created pronoun-poems about the different people or couples that were discussed. Throughout Chapter 4, I have used pronoun-poems (Gilligan, 2015, Woodcock, 2016) to enrich the reporting of my research. This story is told from my perspective and includes poems that emerged from data collected from participants. Chapter 4 summarizes the primary themes that emerged from this analysis.

Positionality

Throughout my study, my positionality took multiple forms, as granddaughter, daughter, sister, cousin, niece, wife, and mother. I primarily hold an indigenous insider perspective because

I was socialized in the community I am studying (Banks, 1998; Chavez, 2008) and am perceived by my participants as part of the group and worthy of receiving and sharing their stories. However, some topics in my interviews were stories I knew little about and shifted my positionality to an indigenous-outsider perspective. Indigenous-outsider perspective describes those who were socialized in the community (e.g., I am part of the family) but have assimilated into an outsider culture (e.g., I am an academic and not of the same generation of some of the storytellers; Banks, 1998; Chavez, 2008). I acknowledge now the potential power dynamic between myself and my participants. That being said, the majority of my participants spoke about father and grandfather based on their relationship to them – "my brother" or "my cousin" rather than "your father" and "Everett" or "Daddy" rather than "your grandfather." To some extent, I believe that my participants were so engrossed in telling their childhood stories that they forgot about my personal relationship with them.

Trustworthiness

In qualitative research, trustworthiness refers to the extent to which the findings are credible, transferable, dependable, and confirmable (Lincoln & Guba, 1985). Credibility is confidence that the findings represent the "truth" while transferability means that the findings could be applied in different contexts. Dependability is similar to reliability in quantitative studies and assesses the degree to which the findings are consistent and could hold up to repeated examination. Finally, confirmability relates to neutral and objective findings. This component is complicated by my positionality in this study and the autoethnographic method.

Autoethnography does not intend to be objective or neutral. As a method, it is designed to be subjective and personal. For this study, I worked to ensure credibility, transferability,

dependability, and confirmability through prolonged engagement, triangulation, peer review, and reflexivity.

This study included data collection through interviews with four generations in my family over a period of about two months (December 2019 and January 2020). Though this effort may not be intensive enough to be considered prolonged engagement, the collected data provided measures of internal consistency. For example, I collected data from my family at over 13 points (12 participants and my own memories and reflections) and reviewed five oral history interviews conducted by historians for the New City Library Oral History Collection. There was cross-interview consistency in the stories participants shared. These multiple methods were triangulated as a measure of internal consistency. Finally, I used peer review with critical friends, including other doctoral students and faculty, to help me imagine how my analyses and conclusions might be wrong. My critical friends were useful in pushing me to consider how I might be wrong and helpful at ruling out potential "plausible alternatives and threats to interpretation and explanation" (Maxwell, 2013, p. 281). I used the aforementioned researcher journal to review the potential biases and perceptions I documented over the course of the study.

Analyses and Reporting Ethics

I was attentive to the ethical issues of autoethnography (Wall, 2016), using member checks to confirm what participants shared with me. I shared a transcript of the interview with participants and confirmed the accuracy of the transcript with them during a subsequent phone conversation. These conversations also were an opportunity for the participant to clarify or expand on any of their statements. I took notes to record what was said. I also confirmed my participants' willingness to review drafts of analyses at the start of the study. Member checks were not only conducted with the participants. I also contacted the people the stories were about

to confirm the story could be shared. Pseudonyms were used throughout this dissertation to protect the privacy of individual respondents. During the consent process, I also informed my participants that despite using a pseudonym, their identity may be interpretable. Finally, there is a relational ethic to using the Listening Guide, requiring that the analyses and reporting are attentive to the multiple voices within the data. In addition, I used my critical friends to keep me focused on themes emerging from the data rather than solely my own opinions or preconceived ideas (Wall, 2016).

The following chapter presents the main themes of this study regarding the construction and transmission of egalitarianism over generations, including relevant poems that emerged from the data collected in this study.

CHAPTER 4: FINDINGS

This chapter presents the findings from my data analyses and my reflections to answer the research question that guides this study: How did the social-historical context, as filtered through previous generations, influence my conception and practice of egalitarian parenting? This section presents the themes that emerged from the analyses about the transmission of egalitarian parenting over generations.

Themes

The collected data was analyzed at the individual and generational levels. The following eight themes emerged from the data: social-historical context, education, husband support, flexible gender roles – female, flexible gender roles – male, partnering, balanced parenting, and intergenerational parenting similarity. Each theme is discussed below along with accompanying pronoun-poems to elucidate the finding.

Social-Historical Context

In the case of our three generations, my family had varying experiences with discrimination, particularly during the Civil Rights Movement. Participants in Generations 3 and 4 were not alive at this time. Perhaps because Nyack was so far from the Jim Crow south, the Black people in my family did not experience race-related discrimination. The following three pronoun-poems share statements from participants in Generation 2 about their communities:

It was very rare in Nyack when I felt like I was a Black man.

I never felt any racial implications.

I liked Nyack.

I don't know how to describe [the Civil Rights Movement].

I saw it happening.

I read about it.

It would be on TV.

I never felt like I was a victim of it.

I was definitely aware of it.

I saw people participating.

I knew and understood why.

It was something on the side

Not a direct impact on what I was doing.

I never felt I should have or should not be involved in it.

You could see it wore on my mother and my father.

- Generation 2 – Age 17 in 1965

I didn't see color as an issue growing up in Nyack.

I didn't even really notice that [my best friend] was White until someone else mentioned it.

I'm sure it wasn't entirely that way.

I really didn't and many others of my peers didn't see color as an issue.

I wasn't really impacted at all by desegregation.

I saw on the news.

I heard about riots in Nyack where I was raised.

I never witnessed anything.

- Generation 2 – Age 7 in 1965

I didn't know anything about hatred and all of that stuff because of our environment.

I didn't know anything about it.

I was too young to really know anything about what was going on.

I couldn't tell you about it.

I don't believe we had any problems.

- Generation 2 – Age 5 in 1965

While these participants may not have had any problems with race-related discrimination during the time of the Civil Rights Movement in the United States, many other Black Americans, particularly in the southern United States did (House, 2010; Johnson, 1954; McKee, 1993; Moody, 1968; White, 1999). The Civil Rights Movement in the South often featured violent reactions to Black activism. I selected participants 'I-statements' and provided their ages to show a range of awareness to the movement within one generation of brothers. In addition, I provided the participants' approximate age in 1965 as a means of situating their developmental age in context. I selected the year 1965 because that was the year of the National Head Start Day event that their mother had attended, as well as the year the United States entered the Vietnam War, the assassination of Malcolm X, and the march on the Edmund Pettus Bridge in Selma, AL resulting in President Johnson's signing of the Voting Rights Act of 1965. This context is important to

understanding why it was so significant that a Black woman was invited to the White House this year, as this was a time of particular societal unrest for Black people in the United States consisting of much discrimination in general.

Others in my family did experience race-related discrimination, especially once they left their communities. My father enlisted in the United States Army after high school. He was selected for Officer's Candidate School, graduated from Fort Benning, and was sent to Alabama. There, he experienced racism, perhaps for the first time. This was a turning point in his life.

He often didn't want to talk about civil rights.

He would act like he didn't believe it was necessary in a way.

I don't know if that came from where he grew up.

He always felt that he had opportunity.

I believed in civil rights a little bit more than he did.

He didn't talk about...

He was in the Army.

He told me this many years later.

He commanding officer had called him the N-word.

He couldn't really talk about it for a long time.

- About Generation 2 – Age 18 in 1965

He was given a pistol to protect himself.

He was in the military in the South.

He was in charge of an honor guard.

They would receive the bodies that were coming from Vietnam.

They would go and present the flags to the families and serve at the grave side.

He was provided with a pistol.

People in the South didn't want to see a Black man in a uniform.

Growing up in Nyack, he had privilege.

Going to Alabama, he was just...

He wasn't expecting.

- About Generation 2 – Age 18 in 1965

Others in my family also discussed racism they experienced after leaving the Nyack area. My great-aunt describes:

I left Sparkill.

I left this house and went to Orangeburg.

People were going out of New York City,

Running away from Black people,

Ran right into me and my children.

My oldest daughter had a very hard time with the N-word.

We never heard...

I'd never heard it.

I was never called that.

It was tough on my kids.

- Generation 1 – Age 26 in 1965

Albany was a different community than Nyack. The State capital was racially diverse though somewhat segregated by race. Growing up in Albany, my mother reported feeling an increased tension in the community around the time of the Civil Rights Movement. She recalled:

My neighborhood was mostly White.

There were Black families.

My mom was friends with one of the women.

I had been in their home before.

We actually had to have a curfew in the city.

You had to be in your home.

I forget what time it was.

My dad was a person who believed in treating everyone fairly and the same.

He would often make that statement:

'I treat everyone the same.'

He was like that.

Young black boys would come.

Sit on our porch.

Wait for my father to get home to take them to baseball.

I can remember one neighbor yelling at our house.

- Generation 2 – Age 9 in 1965

Across Generations 1 and 2, my family had varying experiences during the Civil Rights Movement, with some family members personally not affected and others directly impacted by discrimination. I had a similar experience with the shocking revelation that a new environment was not as inclusive as my hometown.

I was born the first biracial daughter of a Black man and white woman in a college town in the northeast. My White mother is the daughter of a second-generation Italian immigrant and a French-Canadian/ English colonist whose ancestors were early settlers of Massachusetts in the mid-1600s. My Black father is the son of Black parents – one whose ancestors were free Blacks

from Skunk Hollow descended from a merchant on a Dutch West India Company trading ship, the *Bruynvis*, that sailed to New Netherland in 1626 and another descended from Virginian slaves. Interracial couples were not an anomaly in my hometown; multiple students in my school had parents of different races. My friends were singularly White, Black, or Hispanic or some mixture of these. I have always embraced my multi-ethnic heritage, which includes all of these viewpoints – the immigrant one, the Mayflower one, the free Black one, the enslaved Black one. Growing up in a racially diverse, progressive, inclusive bubble, I never had to pick a side. Our bubble was round; there was no side. It wasn't until I left this bubble that I understood the need to choose.

I went to college in a southeastern city still holding on to the vestiges of the confederacy; the Stars and Bars flew atop the statehouse only a few blocks from campus. Though I never experienced any explicit racism, I was subjected to the microaggressions of White people who had limited exposure to people of color. A friend's parents were stunned by how articulate I was and my roommate who "didn't see color" openly mused about why Mariah Carey was suddenly working with "so many niggers." I have never felt more Black. No matter how diluted my blackness, in the south, I would be Black and so I decided I would be. Nine hundred miles from my bubble, I built a new one. This new bubble was still progressive and inclusive but it was comprised almost entirely of Black people. All of my friends were Black. I joined a Black sorority. I immersed myself in the writings of Black women writers — Toni Morrison, Zora Neale Hurston, Gloria Naylor, Ntozake Shange, and Alice Walker — whose stories reflected my own. I found myself in Sula, Janie, and the Lady in Yellow and through these characters I felt connected to the people in my new bubble. I grew into a Black woman with a Black husband and Black sons connected to the Black community. In a pronoun-poem, I reflected:

Growing up in Albany, I never felt other than.

I realized...

I was the only...

I never felt.

Going to school in the South, I felt and was Black,

I built...

I found...

I grew.

- Generation 3

There may be gender-based perspectives that need to be taken into consideration with regard to this theme. Family members who reported feeling the strain of their communities or experienced inner tension were all women, while those who reported not being affected by racism were men.

Education

The pursuit of postsecondary education was a prominent theme of the participants with my family. Each respondent mentioned higher education in their interview, particularly regarding the value of education. Many women in my family, from Generations 1, 2, and 3, (my grandmother, great-aunt, mother, and myself) attended college after their children had begun formal schooling. The following generational pronoun-poems share statements from participants about the role of higher education in the lives of these women in my family.

I was supposed to just go from high school to college.

My husband convinced me I needed to get married.

I got married right out of high school.

I went to college after having five children.

I went to college and decided.

I got work to go to college.

I went to Rockland Community College at night.

I got it.

I went to St. Thomas Aquinas at night.

I have a computer science degree.

- Generation 1

My mother never gave up on getting education.

She would take courses.

She spent a lot of time going through...

If my mom saw some course she felt would help her advance her educational goal, She was participating in it.

I went to school

Then, she went and got her...

She completed her formal education.

She's got a couple of degrees from Rockland Community College

She got a couple of degrees.

My mother would have done her PhD if she had enough time.

She definitely always wanted to have her degree.

- Generation 1

She was going to Maria College to become nurse.

Mom was in school.

She was at class a lot.

She wasn't home.

She went to school.

She had three children and worked a full-time job.

She was working and going to school at the same time.

She could do what she needed to do.

- Generation 2

She has class.

She might not be home in time.

My mom will probably be working on her dissertation.

She gets the chance.

- Generation 3

Across generations, there is similarity in how higher education was pursued by women in my family. We all were married mothers at the time of our enrollment in college though differences exist with regard to the level of education. My grandmother and great-aunt achieved Associate's degrees, my great-aunt and mother achieved Bachelor's degrees, and I achieved post-graduate degrees after we each became mothers. In addition, we all were working at the time of our enrollment in college. In my family, navigating these simultaneous roles – wife, mother, worker, and student – required the coordination of multiple responsibilities and the assistance of multiple others.

Husband Support

In the case of our three generations, the most frequent supporter of our higher education pursuits were our husbands. As married working mothers, we were not the sole financial providers. All of our husbands worked; some to excess. There are some differences across generations about the intensity of work outside the home. In this study, my grandfather is frequently described as fiercely dedicated to work, doing whatever he could to provide a good life for his family. Multiple participants told the story about how my grandmother would load up their sons in the car and visit my grandfather at work, because the family would not see him otherwise. Representing Generation 1, my great-uncle also was a supportive husband. My father had a different approach to work but was supportive in his own right. The males in my family representing Generation 3, including my husband, appear to have achieved greater balance between work and emotional and caring support. The following generational pronoun-poems share statements about Generation 1's support of women in my family:

My dad was working the 3 to 11 shift.

We would go down in the evening time,

Take him his dinner,

And sit out by the Hudson River.

We would go to the water and throw skip rocks.

My mom and dad would talk.

My brothers and I would be walking in the water, having a good time.

My dad would take us in the mill every now and then ...

We would go in to get him...

- Generation 1

My dad went to work.

He worked two jobs.

His mission was to provide for his family.

He worked different shifts.

If he wasn't working a day shift at the mill, then he was driving a truck.

He was always leaving, going to work, whether to drive or work at the factory.

He did everything he could with the multiple jobs.

My dad was still going to work two or three jobs - full time, part time, overtime at the full time.

My dad was a...

supervisor,

workaholic,

servant.

His work ethic was unbelievable.

He wouldn't not work when it was good money.

- Generation 1

Her husband was very good to her.

He came home from work and put his check up on the counter for her to take care of.

He had no issues about that either.

He never cared that she was doing all this stuff.

He let her do whatever she wanted.

He knew her.

He knew she wasn't gonna do anything wrong.

He never complained about anything that she did.

He would give it to her.

He would bend to her will.

He would do for her.

He went.

He was right there.

He was involved.

He would always tell her, 'Yeah babe, you can do it.'

He supported her and everything she went out to do.

He was right there - 100% supporting her, encouraging her.

Dad was supportive of her work.

Dad would support it.

My father would do whatever he could to help my mother's dreams come true.

- Generation 1

He might say:

This is what you're going to do?

I'll make sure all of this is taken care of.

That's what you want to do?

You don't have to worry about the house.

You don't have to worry about anything here.

I got this.

You do what you need to do.

- Generation 1

While men in Generation 1 were primarily providers offering tangible support (e.g., financial assistance, material goods, or services) to their spouses, Generations 2 and 3 primarily offered emotional and esteem support (Ko, Wang, Xu, 2013) to their wives, often taking on non-traditional gender role responsibilities. Over the course of my data collection, I learned that my

father was a childhood friend of Daniel Friedan, whose mother Betty wrote The Feminine Mystique (Friedan, 1963) while living in Rockland County. I wonder if my grandmother knew her and if that book had influenced her in any way. I speculate that my father became a feminist through the environmental influence of Mrs. Friedan, his mother, and aunts. The following generational pronoun-poems share statements about how men from Generations 2 and 3 have supported women.

Daddy would literally kick us out of the house just so we didn't make any noise.

He was very protective of her.

Daddy would send us out of the house so she could rest.

He demanded that we respect her.

- Generation 2

We're both feminists who appreciate the dedication of a hardworking man.

That dedication to their work.

That dedication to provide.

They're not just doing it because it looks good to do it.

They're doing it because of the sacrifice in service for the family.

There is a partnership.

'I'm doing this for us.'

I think every male person,

Our dad.

Our grandfather,

Your husband.

My husband.

These are men who are fiercely dedicated to family.

That allows for us to feel vulnerable at times.

We don't have to be so hard-edged all the time.

We can be vulnerable with our spouses because they understand our partnership.

They understand our worth.

- Generation 3

I leave first.

I come home first.

I'm home.

I am present in the house.

He wakes me up.

He makes me breakfast.

I'll drive them to school.

I leave work about one o'clock pm to pick up children.

I pick them up pretty much every day.

I'm getting them at different times, but I still pick them up.

I do my best.

We figure it out.

- Generation 3

Across generations, husbands supported their wives endeavors by providing care for the home or children. From G1 to G3, there is increasing specificity about the type of care that was provided. This is likely due to the proximity of these activities, as participants representing G3 are currently providing these supports and participants from G4 are receiving them. G3 and G4 experiences are current while the husbands described by G1 and G2 are deceased and provided these supports 30-55 years ago.

Flexible Gender Roles - Female

Across three generations, respondents discussed the role of women in society. Generation 1 shared strong feelings about equal rights for women and advocated for themselves, particularly during the 1960s-1970s when women were increasing their presence in the workforce. In Generations 2 and 3, participants were mixed in their perceptions about who they would be. The following generational pronoun-poems share statements from participants about the role of women:

She was a driving force, rebellious, and brave.

My mother was friends with a woman who "burned her bra."

She supported that.

My mom was very involved.

She often complained about unequal pay.

She joined different organizations to combat unequal pay for women at factories.

She wasn't afraid to go up against something that she didn't feel was correct or right.

She had a very big heart.

She cared.

She ran, went, traveled.

She was in charge.

She was recognized.

Generation 1

Women give up their dreams because of children.

They stopped life because of children.

My mom might've done that.

She wanted to do.

She had the opportunity.

She was able to do.

She says, 'I'm outta here.'

She began to travel.

She's been to Africa.

She's been to a lot of foreign lands.

She went to England.

She's been there.

She was free-spirited.

She wanted to do.

She did.

Generation 1

Mommy told him.

I told my husband

'I gotta get out of here.'

'I gotta do something.'

I told my husband.

I had to talk to somebody that stood taller than...

I went for an interview for a job.

I made my husband quit one of his jobs.

I would go to work then straight to college.

My husband had the children.

He was a good cook.

He didn't mind.

He liked to cook.

He was from the South, and he didn't mind.

Generation 1

I was the only female, Black and all White men.

I wrote a program.

I got it from the mathematical book.

They were so proud, 'Blah, blah, blah.'

I became project manager.

One White man wanted that job...

They gave it to me.

Again, only Black female in the room.

That went on for years.

- Generation 1

I went to work.

I was just a clerk.

I was in a room with nine other White women doing paperwork out of the computer room.

I decided that we weren't being treated fairly.

Every time you see a man, he's in the computer room bypassing 10 women.

I went to my supervisor and said...

'I want to discuss some discrimination.'

He thought I was talking about being Black.

I was talking about being female versus the males.

I said...'You had 10 women

You never give them an upgrade in pay or job position.'

I said, 'I want to go to college...

I understand that some people come have been sent to college after they've been hired.'

He said, 'Oh, no problem.'

He threw the papers up on the desk.

He said, 'What would you like to take?'

- Generation 1

I felt torn to be honest with you.

I had never really seen my mother work outside the home.

I very much was interested in being a mother.

I had my own children.

I was working.

I had them.

I wanted to be home and raising my children.

I felt to be a whole person...

I had to be working and had to be achieving.

I had to do both of those things.

I felt...

I still was picking traditional roles.

I first became a teacher.

I became a nurse.

I was still in that mindset.

I could have been...

I could have been a whatever.

- Generation 2

I was elementary age.

Dad played the Mom role.

Mom played the Dad role.

I got in middle school and high school.

Mom was more present.

Mom was home.

Mom would be home cooking dinner.

- Generation 2

I always knew.

I would work.

I saw.

I remember Baby Boom.

The 80s were filled with films about working moms.

I would have a career.

I didn't know what.

I would have it all.

I believed that.

Generation 3

I think that's what weighs on me.

My husband is the major breadwinner.

I have this feminist mentality.

'Women can do everything.'

I can't.

I need this person.

- Generation 3

For Generation 1, equality and inclusion was focused primarily on gender. For Generations 2 and 3, respondents discussed societal pressure about the need to be both working, independent women and being mothers.

Flexible Gender Roles - Male

Men from Generation 2 and 3 shared that they had taken on non-traditional parenting responsibilities. An uncle took a part-time job at the library so he could bring his daughters to work with him. A cousin is a stay-at-home dad. My husband and I have negotiated our work schedules such that I am home in the morning to get the kids ready for school and he picks them up from school and takes care of after-school responsibilities. The following generational pronoun-poems share statements from participants about the flexible gender roles of men in my family:

He'd be there.

He was always there.

He wanted to be there.

He wanted to be present.

He was just always around.

He was very involved.
He was very much a nurturer.
He was like a helicopter parent.
Mom was the breadwinner.
He didn't want to just be a provider.
He raised you a little differently.

- Generation 2

I am present in the house.

I do my best to make the kids breakfast in the morning before they go to school.

I'll drive them to school.

I pick them up pretty much every day.

I'm getting them at different times.

I still pick them up.

I let them be involved in sports and activities that would fall on the weekend.

I still make them cut the grass and gather the wood when the trees break.

- Generation 3

Between Generations 2 and 3, men in my family began to take on nontraditional gender roles. These men were involved in caring and nurturing behaviors, regularly maintaining the household and children. Growing up, my nuclear family consisted of two working heterosexual married parents earning a middle-class income and three close-in-age children. People would likely describe my father as very involved despite the fact that he was holding down a full-time job. My Black father was abnormal, when compared to the stereotypes about Black dads — absent, aggressive, and unemployed soldiers of the street life. My father was none of these things. He wasn't holding down the block; he was coaching little league. He was a nurturer and a thoughtful and measured authority. He was the type of man that could break your pubescent spirit and shame you with six words, "I am so disappointed in you." When asked for his opinion, he would encourage you to voice your ideas and helped you puzzle through your next steps. He listened, empowered, and trusted. He modeled fairness and rewarded effort regardless of success. I remember him as a feminist.

My father was actively involved in our care. He made dinner, helped with homework and projects, and chauffeured us to practice and Girl Scouts. While my mother attended college in the evening, my father was our primary caregiver. I would say that his greatest accomplishment was being a father. He thrived in these roles as mentor and coach. He was a dad to many of my friends who did not have relationships (or good relationships) with their fathers. I could tell multiple stories about high school friends who, years after high school, sought out my father to borrow a tie for a job interview or discuss an issue he was having with a girlfriend. Yet, I do not think my father ever fully understood his impact on the young people whose lives he touched. I believe he felt inadequate because he was not as financially successful as one of his younger brothers (an MIT graduate who retired as an IBM executive). My father's success is one not measured in dollars but in the number of young men he inspired and the number of young women who knew how to recognize a "good man" by his example.

I was lucky to find my own "good man" whose selfless care for me and our sons has made this dissertation possible. In his own words, he says:

I'm a very good father... a very, very good father.

Being selfless is central to being a good dad.

It's something that every father has and then develops.

He fine tunes his ability to be selfless.

Makes it more pronounced.

Maybe it's something that he doesn't readily recognize.

Maybe it's something that he feels he shouldn't be.

I think it's very rewarding.

Not just your children,

Your spouse,

Your household,

Whatever in your environment needs to be maintained.

It's maintained to the best of your ability.

A good dad is that person that fulfills every possible role or responsibility.

Whatever duty.

It means to participate unconditionally.

Everything involves family, whether you like it or not.

- Generation 3

Men in my family took on parenting and household responsibilities above and beyond what traditional fathers might.

Partnering

Across generations, participants shared the similar memories about our parents and their relationships with their spouses. Participants shared stories about the communication – verbal and non-verbal - between partners. The following generational pronoun-poems share statements from participants about the ways couples interacted in my family:

They were committed to each other.

They communicated strangely.

They would argue.

I wouldn't like.

They would argue.

I didn't like it.

They always kissed afterwards and hugged.

My father would say something like, 'Do you want to go upstairs?'

My mother would say, 'I'm not going to kiss you later.'

It was very strange to me how people could argue with each other and still love each other.

- Generation 1

Daddy would tease Mom a lot.

He 'messed' with her.

Even the arguments were a little on the sarcastic side.

He would try to make her laugh.

They were goofy together.

They were very loving and playful.

They reminded me of Grandma and Grandpa.

They would do their little bickering, just a little back and forth.

I just thought it was cute.

I figured out what was happening.

I just put a smile on my face, 'Oh wow, they're up to it again.'

I really appreciate them.

They were an inspiration to me.

I saw love all the time.

I always saw love.

I saw together in unity in your parents

They had a very happy marriage.

I can remember them dancing around in the kitchen.

- Generation 2

I wanted a relationship like that.

They could talk to each other like people.

They were very vulnerable with each other.

That's what made them have a strong relationship.

- Generation 2

I do get the PDA from them.

'Hey, stop touching my butt!'

Our parents did that.

My husband is always copping a feel.

My husband is always pinching my butt.

Obnoxiously enough.

I definitely got that from our parents.

My children know, 'Mommy and Daddy love each other.'

The love for each other is so strong

It consumes the rest of the household.

That kind of love for each other will trickle down to the children.

- Generation 3

My mom and dad as a couple are cool.

They love each other.

My dad has to go to work

My mom is up, so she can say bye to him.

They also kiss a lot.

They love each other.

They love their kids.

- Generation 3

My grandparents were an interesting couple. They certainly loved each other...but it was a love that was close to exasperation sometimes. My grandmother would simply be fed up with my grandfather's antics. Their playful banter, my cousin and I realized recently, was like foreplay usually resulting in a question about "later." I remember their slight bickering, but not the coy references to the bedroom. I may have been too young to pick up on these. My parents behaved similarly. Their commitment to each other extended beyond maintaining the household. I remember them as true partners. Their love and desire for one another was evident; their affection now is something I can look back on fondly. As a teenager I would roll my eyes; but as

an adult with my own marriage, I realize that I am equally as affectionate with my husband. We joke, bicker, and frequently get caught kissing in the kitchen by our kids.

Across generations, there is similarity in the way couples displayed physical intimacy in front of others, their children in particular. For me, I generally feel unaware that my children may be watching. I feel that the public displays of affection between my husband and I are evidence of our connection. I would suspect that this was true of my parents and grandparents as well. This connection may also explain couples' commitment to support one another.

My parent's song features the lyrics "I just want someone that I can talk to. I want you just the way you are" (Joel, 1977). These words were true of them. What cuts across as I think about my grandparents, parents' and my own marriage, is a sense of friendship. Within these three sets of couples, there is tremendous respect for the other person. For me, one of the things that facilitates egalitarianism is respect for the other partner's humanity. This does not mean a simple yielding to give someone their way. Rather, it is an understanding and recognition of the other person's needs and wants, and providing care – be it reassurance, advice, tangible support, or physical comfort.

Balanced Parenting

Over generations, participants shared stories about the parenting they had received and practiced. Many participants reported that their parenting emphasized respect, high expectations, rules, and explanation. The following generational pronoun-poems share statements from participants about the parenting styles of their parents or grandparents:

Grandma always listened to me.

She always wanted to hear my side.

She never believed I did things for no reason.

She was a reasoner.

She would explain things to us.

Help us to understand from cause and effect.

If I do this, this is the result.

She would say, 'I want to know why you did it...'

She would ask, 'Is that really a reason?'

She could see it in my face.

She said, 'You know what? I believe you. You're not in trouble.'

She was the first adult that really listened to me and cared to hear what I had to say.

Generation 1

I thought they were peculiar parents.

I had to have a talk with them.

I used to hear you call your dad, 'Dave.'

I asked Dave.

I said, 'Why do you allow them to call you Dave?'

And he said, 'That's my name.'

I said, 'Yeah, but you're the parent and you're older and they should call you Mom or Dad.'

And he asked me, 'Why?'

I said, 'Because, it's respectful.'

And he replied, "Do you think my kids are disrespectful to me?"

Your dad would say:

"My children are people.

They're just little people and that's how I treat them.

That's how I talk to them, as if they're just small people."

That struck a chord with me.

It influenced how I raised my children also...

To respect them as people.

I didn't understand that.

Dave respected even children.

As if they had all the rights of an adult.

It was difficult to understand that.

I benefited from that.

- Generation 2

I had a great foundation.

I came from good parents who had good parents.

He was my coach.

He was my best friend.

He taught me a lot in life.

He was always very easy to talk to.

He was always accessible.

He was gonna be there.

He would help me.

He always treated me like I was a grownup,

Even when I was a kid.

Mom too.

I always felt like they respected me as a person.

They didn't treat me like they were above me.

They treated me like I was my own individual

They respected me and allowed me to grow into who I wanted to be Rather than who they wanted me to be.

Generation 3

I need to interact with my kids. I need to speak to my kids. I need to understand. I need to start tweaking now. I talk to my kids. I'm going to understand. I'll learn that.

I'll be able to cater to them better.

- Generation 3

Men in my family were more often balanced parents than strict autocrats. Their comments reflect nurturing, listening, considerations for others' feelings, and providing their children opportunities for responsibility.

Intergenerational Parenting Similarity

Across generations, participants shared stories about similarities between their parenting styles and those of their parents. The following generational pronoun-poems share statements from participants about what parenting styles or aspects were passed down to them:

I did a lot of the same things that my mother had done.

I used to read to my children,

Buy them books,

Take them to the library.

I tried to take them to things in the community.

I tried to be more involved with the school.

That's something my mom had never really done.

- Generation 2

I customized it.

I pulled from my experience.

I customized it.

Including all their different styles.

Some things I wasn't going to do.

Some things I definitely was going to do.

I did not want to be a disciplinarian like my father.

I pulled from each one what I liked,

Incorporated it into raising my three daughters.

I pulled from all those resources of family.

I did my best.

Generation 2

He loved his dad.

He felt less close to his father.

His father was absent a lot when he was growing up.

He was working two full time jobs.

He wanted his dad to be there.

He was on the team.

His father didn't come.

He had missed out.

David became a father

He was very conflicted.

He wanted to be there.

He wanted to be present.

He didn't want to just be a provider.

Later, he was able to spend more time with his father.

He was able to come to terms with things.

- Generation 2

You're a reasoning person.

Your father was a reasoning person.

There was a lot of talking and reasoning out.

I've seen it.

I watched you with Marcus.

I just observed it.

I saw how David was teaching you.

Dave was a reasoning...

A reasoner.

He talked with you all.

I see that you talk with your children.

Marcus at the time.

You got the point across that you wanted to get across.

He understood it.

- Generation 3

I'm very strict.

My grandparents were equally strict.

I think it was a different type of strict.

Because of how times and environment has changed in comparison to 30-35 years ago,

I'm a little more guarded.

I'm a little more the watchful eye.

I'm conscious of what's going on around me.

I always tend to interject just because it's not the same as when I was my son's age.

I could be outside by myself and be in our backyard for endless hours.

I think we can do that today, but you find yourself kind of looking over your shoulder.

Generation 3

I am the opposite of my father. I chose to improve. I'm exactly how my grandfather was. I modeled it after my childhood.

Generation 3

Across generations, participants shared what they had learned from the people who raised them. In most cases, that person was a parent, and in some cases, it was a grandparent. Participants remarked on their intentions to compensate for undesirable parenting practices and model effective parenting practices they had observed over time.

These eight themes were further reorganized into the following three meta-themes: parental socialization, partnering socialization, and intersectional socialization. Socialization refers to the ongoing way that values, behaviors, and expectations are constructed and transmitted over generations through messages, behavioral reinforcements, relational approaches, and modeling (Rivas-Drake, Hughes, & Way, 2009). Within my family, I have been socialized as a worker, spouse, parent, and human. These multiple identities are interconnected.

Parental socialization refers to the values, behaviors, and expectations about what it means to be a parent that were communicated to me and others in my family. This includes mothering, fathering, and co-parenting expectations. This meta-theme includes the themes of balanced parenting and intergenerational parenting similarity. Partnering socialization refers to how men and women in my family have nontraditional gender-based parenting roles and share responsibilities. This meta-theme includes the themes of flexible gender roles (male and female), husband support, and partnering. Intersectional socialization refers to the development of my identity as a Black woman and includes the themes of social-historical context, flexible gender

roles (female), and education. These three meta-themes are discussed in conversation with the broader literature in Chapter 5.

CHAPTER 5: DISCUSSION

This autoethnographic study is a personal account of how my family constructed and transmitted parenting style over generations within specific historical time and space. This account reflects the self-analysis and self-reflection of a working Black mother who is pursuing a doctoral degree. Despite all that I had read about the vulnerability required to craft an autoethnography, I was unprepared to take on this task. The difficulty of putting "pen to paper" so to speak has been exceedingly more challenging than I would have ever expected. At the same time, there is therapy in this work; it is somehow cathartic and heartbreaking. There have been many tears, from my eyes and those of my participants; I have been undone and restored. That being said, this method speaks to me. Much like autoethnography as a method, I would describe myself as "vulnerable, rebellious, and creative" (Ellis & Bochner, 2006, p. 43). The findings may provide an example into how intergenerational ties in a Black family and context influence individual development.

Summary of the Study

This study presented an autobiographical account of the intergenerational transmission of values and practices. The intention was to show how egalitarian parenting was constructed and transmitted over generations – from Generation 1 to Generation 2, from Generations 2 to 3, and prospectively from Generation 3 to 4. ⁴ My study was designed to answer the question: How did

⁴ Overall, little usable data was collected from Generation 4. At the time of their interviews, my sons were ages seven and 12 and did not share much regarding how they might one day be parents. Their responses are integrated in pronoun poems developed in Chapter 4.

the social-historical context, as filtered through previous generations, influence my conception and practice of egalitarian parenting?

Life course theory theoretically framed my study, as it relates to the principles of *linked lives* and *historical time and place*. African philosophical concepts of sasa and zamani also supported the theoretical framework. I used autoethnography, a qualitative research approach, to investigate my research questions. Autoethnography is a study of one's own culture and focuses on what is internal to the researcher. It narrates the researcher's lived experiences and is written using a first-person voice.

This study was conducted to add underrepresented voices to the family science literature, particularly Black families and their strengths-based processes. I occupied multiple spaces in this study. I was simultaneously the researcher collecting the stories of four generations of my family and a participant, providing my own experiences and stories. In a number of interviews, I was rehearing stories that I had originally heard as a child, understood with an immature mind, and had new perspectives and considerations hearing them again as a wife, mother, and scholar. For this study, I situated myself inside the culture of my family. I reflected, listened, observed and analyzed the collected data to determine how the social-historical context influence my conceptions and practice of egalitarian parenting. The collected data consisted of 17 oral history interviews and audio-taped recordings of my own reflections, totaling over 11 hours of narration. My findings identified a number of themes that were shared across generations, including social-historical context, education, husband support, flexible female and male gender roles, partnering, balanced parenting, and intergenerational transmission. These themes were combined into three overarching categories – parental socialization, partnering socialization, intersectional

socialization. In the section below, I, first respond to the research question, then discuss the three overarching categories within existing literature.

Intergenerational Transmission of Egalitarian Parenting

My conception and practice of egalitarian parenting was influenced by the socialhistorical context primarily through an active pursuit of what I saw was possible. Participants shared the direct impact of social movements and societal context on their development and drive for equity and inclusion. For example, my great-aunt discussed her self-advocacy with regard to being female in a male-dominated workforce during second wave Feminism and the Civil Rights Movement. My grandmother advocated for equal pay for women in factories during World War II. She was heavily involved in her community, lending a hand to the Civil Air Patrol, the American Red Cross, her church and youth organizations, and the launching of Head Start in 1965. My mother shared her inner turmoil at the societal pressure to be more than simply a mother in the 1980s when women were given accolades for managing work and family. Family members reported my father's unconventional approach toward parenting, trading in authoritarian values of mutual respect for strict obedience. Further, my father rejected the expectation that he would be a provider like his father and instead worked nontraditional hours in order to be present. My father was a nurturer when men were expected to be instrumental providers. Reflecting on my experiences and understanding allowed me to integrate other principals of life course theory.

Reflecting on my individual development helped me to realize that I actively sought to replicate my grandmother's traits and accomplishments. The life course theory principal of agency became particularly important as I continued with data collection. I stated:

For me, I intentionally followed my grandmother's example. For my siblings who also represent Generation 3 in my study, that transmission was not present. This could be because of individual differences and enacting something over time... In reflection, I think that I intentionally replicated my grandmother. As a Black woman – the most prominent Black woman in my life - she was a model for me. I have organized my family to replicate hers. I think the idea of *agency* and intergenerational transmission is interesting, especially the extent to which you can choose to value something or not. I chose to value egalitarianism by my parents and grandparents example; those were not salient values for my siblings.

Growing up in Albany, I had one Black female teacher in elementary school and one in middle school. There were few Black women in my life, with the exception of my grandmother and a handful of aunts who did not live in my town. I did not have regular interactions with any adult Black women. While I did not feel that I was treated differently in school by classmates or teachers, I generally felt uncomfortable around Black girls. I had been accused as an adolescent of "talking white" or liking only White boys. My grandmother represented something for me. She was who I could be. A fast-driving, independent, brave and determined, community-focused advocate who loved her family, was in-tune to the needs of others, and committed to lifting up everyone.

I did not realize until this study that I actively replicated her. Over time, I have noticed our similarities but failed to recognize my intentional imitation of her. I have marveled at the similarity of my husband and grandfather (her husband) whose birthdays were only days apart, like ours. I would blame our likenesses (ours and theirs) on astrology – the Aries and Libra pairings that were cosmically betrothed. My grandfather was the ultimate provider, working full-

time, part-time, and over-time to earn income for his family. After all her sons were school age, my grandmother wanted to go to college, work outside the home, see the world, be active in her church, and provide for her community. My grandfather was committed to making my grandmother's dreams come true. My grandparents demonstrated egalitarianism by supporting and encouraging my grandmother's pursuits, particularly those that were unconventional for women at the time. As a mother with school-age children in 1965, she had the opportunity to attend college, take on community initiatives, and work outside their home.

Throughout my upbringing, I believe that egalitarian partnerships and progressive attitudes on gender roles were prominent across generations. Between Generations 1 (my grandparents) and Generation 3 (myself), egalitarianism was constructed and transmitted. For my parents (Generation 2), an egalitarian approach to parenting may have been unintentional. My father, who did not wish to take on a provider role at the risk of not being present for his children, and my mother, who believed she needed to both care for her children and pursue a career, were well-matched. They both were able to pursue the lives they wanted through their partnership. From their example, I learned that parents did not have to enact strict gender roles. Both my parents were nurturers and providers who alternated playing a lead role in the care of their children. My mother was supported and encouraged to pursue her educational dreams while they were raising school-age children. College educated women were unusual in the 1980s, but I didn't know any other mothers who were going to college while raising school-age children. For me, my mother's educational pursuits were unconventional.

I believe my husband and I (Generation 3) have an egalitarian approach to all our familial and household responsibilities. As working parents who are committed to seeing a job well done, we regularly juggle professional work, parenting, and household maintenance, while supporting

our extended family from time to time. In addition, for nine of the last thirteen years, I have been pursuing graduate education. There is no adherence to traditional gender roles in our family and our negotiation of responsibilities often is not predetermined. Usually the person who is most able or available or flexible to take care of it in the immediate completes the responsibility. Our approach to parenting is egalitarian though it may not be 'equal' in the short-term.

From the data, three meta- themes influenced my development, construction, and practice of egalitarianism, including parental socialization, partnering socialization, and intersectional socialization. These themes can be integrated with the broader literature.

Parental Socialization

Modeling and Compensatory Fathering

Across generations, individuals and couples in my family shared responsibility, coconstructed values, and respected their spouses' needs. Men exhibited 'new fatherhood' (McGill,
2014), demonstrating involvement in the care of their children, engaging in "more equal"
parenting, and interactive and physical nurturing. Men's perceptions of their fathers shape their
relationships and level of involvement with their own children. The potential for father
involvement is framed within two ideas: 1) *Modeling* proposes that men mirror their highly
involved fathers, and 2) *Compensatory* suggests that men are more involved than their own
fathers because of perceived deficient involvement in order to amend what he received as a child
(Habib, 2012). This theory particularly focuses on measures of physical care and responsibility
(Gaunt & Bassi, 2012). Modeling and compensatory fathering processes are based on social
learning theory (Bandura, 1986) and the concept of intergenerational transmission. To a great
extent, Generation 2's fathering behaviors were intentionally focused on compensating for their
experiences with my grandfather who worked all of the time during their childhood and

adolescence. In addition, my husband did not have a relationship with his father and also may be compensating for his absence. My father and husband were intentionally more involved with their children and maintained that involvement over time. For these men, their compensating behaviors were successful.

Balanced Parenting

In this study, men in Generations 2 and 3 displayed a balanced parenting style featuring respect, high expectations, rules, and explanation. Balanced parenting also is known as authoritative parenting. This parenting style is one of three concepts originally developed by Baumrind (1971) and is most frequently associated with positive child outcomes (Carlo, White, Streit, Knight & Zeiders, 2018; Dornbusch, Ritter, Leiderman, Roberts & Fraleigh, 1987; Majumder, 2016). A fourth parenting style, uninvolved, was added later (Maccoby & Martin, 1983). Parenting styles refer to parents' approaches to interaction with their children.

Per Baumrind (1971), *authoritarian* parents are stricter and focused on obedience and control, while *permissive* parents are indulgent and lenient. *Uninvolved* parents are just that; they have detached and unfriendly relationships with their children. *Authoritative* parents are balanced in their interactions with children. Authoritative parents encourage individuality and independence, communicate, have high expectations, and recognize the rights of people (Baumrind, 1971). Men in Generations 2 and 3 of this study primarily demonstrated authoritative parenting as indicated by participants' disclosures about parents' reasonability, communication, and encouragement of independence. This parenting style is considered by researchers to be the gold standard for approaches to parenting. The parenting style of mothers in Generation 2 and 3 was not discussed at length by participants. It may be that differences in parenting style between

parents could impact the transmission of that style from one generation to the next, perhaps with regard to gender. Mothers may transmit parenting style to daughters and fathers to sons.

Over the years, researchers have attempted to identify the parenting style of couples by averaging parents individual scores (Steinberg et al., 1989; Steinberg, Lamborn, Dornbusch, & Darling, 1994), which was not found to be an effective method of determining the couple's parenting approach. Others have tried to calculate couples' parenting styles using a complicated scaling technique (Fletcher, Steinberg, & Sellers, 1999) or robust family parenting style typologies (Maccoby & Martin, 1983). Further, the original conception of parenting styles (Baumrind, 1991) aligned closely to traditional gender roles, where the mother is nurturing and caring (permissive and authoritative parenting styles) and the father is controlling and strict (authoritarian and authoritative). This traditional view of the association of gender roles and parenting styles was not upheld in a more recent study (Simons & Conger, 2007). Fathers also had more authoritative parenting styles in households featuring egalitarian division of childcare (Sabattini & Leaper, 2004). It may be assumed that the communication required to maintain an egalitarian approach to parenting is well-suited to an authoritative parenting style. There are cross-cultural differences (Mowen & Schroeder, 2018; Sorkhabi, 2012; Weis & Toolis, 2010) in the parenting styles leading to positive child and youth outcomes.

Spanish adolescents with indulgent parents (Garcia & Garcia, 2009) showed the most positive outcomes on self-esteem, psychosocial maladjustment, personal competence, and problem behaviors. Though the authoritarian parenting style is generally accepted as an undesirable parenting style leading to more negative child outcomes, negative academic outcomes were not found for Asian adolescents with authoritarian parents (Chao, 2001). There also are differences in how researchers categorized the vigilance of Black mothers living in a

low-income neighborhood vigilance toward their adolescents, with some identifying their parenting style as authoritarian (Kerr, Stattin, Biesecker, & Ferrer-Wreder, 2003) and others interpreting similar mothers as authoritative (Jarrett, 2000). Culture and environment both moderate the degree to which particular parenting styles are effective (Mowen & Schroeder, 2018).

Partnering Socialization

Ubuntu

The analyses revealed that, for individuals in my family across generations, egalitarian parenting, featuring gender role flexibility and support was sustained despite social-historical movements. For members of my family, men and women were willing to enact nontraditional gender-based parenting roles and share responsibilities. Throughout this study, I have come to realize that the concept of egalitarianism does not well describe the partnership between spouses I hoped to convey. The term, *Ubuntu*, is more appropriate. *Ubuntu*, from South African philosophy, refers to interconnectedness, interdependence and shared humanity and involves support, harmony, and responsibility toward a communal achievement (Nussbaum, 2003; Swanson, 2007). Over the course of this study, I learned that, though egalitarian is how my parents and grandparents relationships might have been described, it was really more than that. Ubuntu is the belief in a universal bond of sharing that connects all humanity. It translates to, "I am because we are." I feel that the idea that people support one another in pursuit of the common good makes sense for me. In my family, no one sought to equitably divide labor (e.g., cooking every other day); support was provided based on need and partners aimed to fulfill those needs. Moreover, I believe that spousal partners in my family have strong connections, exhibiting affection and communication.

Intersectional Socialization

Finally, I realize through this study that my intersectional identity has developed over time. Intersectionality refers to the interconnected marginalized identities that create overlapped opportunities for discrimination (Crenshaw, 1989). As a child, I knew I was a girl, but I was raised without seeing race as a systemic barrier. Race was simply not acknowledged nor was it part of my identity. Now, I would tell someone that I am a Black woman with a White mother. As I've aged and especially when I attended college, gender and race became part of my identity. This is consistent with developmental studies about the formation of racial identity (Csizmadia, Brunsma & Cooney, 2012; Neblett, Smalls, Ford, Nguyên & Sellers, 2009), as racial identity begins to solidify during adolescence. I may have established my racial identity a bit later than most. Further, scholars (Csizmadia, Brunsma & Cooney, 2012) have used life course theory to examine the racial identity of multiracial adolescents, including an examination of principles of historical time, human agency, and linked lives. Biracial youth in the Northeastern United States are more likely self-identify as multiracial or biracial than in the South (Brunsma, 2006). Whereas, biracial youth in the Southern U.S. identify more strongly as Black because they believed that others view them as Black (Khanna, 2010). These findings are consistent with my own experience; I became Black in the south.

My grandmother was the closest adult Black woman to me. She modeled for me what it meant to be a worker, spouse, parent, and human. Black and brown grandmothers have been found to support the identity development of biracial children (Chancler, Webb & Miller, 2017) and studies support an intergenerational impact on the ethnic–racial identity of children of color (Huguley, Wang, Vasquez, & Guo, 2019; Neblett et al, 2009; Rivas-Drake, Hughes, & Way, 2009). Before I was born, my grandmother traveled to Kenya and Tanzania with her Seventh

Day Adventist church. Throughout my childhood, she would take many other trips with her church family, a predominately Black congregation. When I visited my grandparents, I was primarily in the company of Black people, including family and friends of my family. I was socialized about Blackness by her example. While we never talked about prejudice or pride explicitly, she normalized what it meant to be Black.

Racial socialization has been shown to influence adolescent racial identity in Black youth (Neblett et al, 2009). My grandmother was a quilter who made replicas of quilts that were used to direct enslaved people escaping Southern work camps. She adorned photo albums and dish detergent bottles with African Kente cloth patterns. She had a copy of The Negro Revolution in her bookshelf, a book that I now have on mine. In addition, my grandmother was a proud woman, who was vocal about perceived inequity and broadcast her capabilities to do things as well as men. I remember stories about her donning a chauffeur's uniform to drive clients to the airport for my grandfather's limousine service. She was empowering. Gender also has been found to impact the relationship between parental racial socialization and adolescent racial identity (Neblett et al, 2009). Daughters may be receiving more frequent and intentional messages about race than sons (Neblett et al, 2009; Sanders Thompson, 1994). In these ways, she socialized me by imparting implicit positive and inclusive messages about race and gender. When we were together, intersectionality was in the room so to speak; she brought it in with her. She made my inclusive childhood bubble more colorful and her presence in my life likely helped me to further develop my identity.

Implications for Practice

I hope that this study encourages others to use autoethnography to contribute their experiences to the larger body of family science literature. Black family processes, across the life

course, should be studied qualitatively. The stories Black families tell about resilience and strength need to be collected and shared to inform a narrative of how Black families operate. Black family interventions also should consider an increased emphasis on ancestors (extended family) and the role they play in family process. This study was an opportunity for me to write against deficit-based conceptions of Black fathers and families, describe the transmission of values over generations, and add my story.

Limitations

This study aimed to collect information about how egalitarian parenting was passed down over generations by relying on individual interviews and singular reflections. Egalitarian refers to fairness and balance in the distribution of tasks between multiple individuals. This study could be improved by focusing on the negotiation between couples related to their construction of egalitarian partnerships. Missing from this study was evidence of the informal and formal negotiation processes couples use (Daly, 2002). My husband and I value time, which we find to be in short supply. We negotiate responsibilities based on expediency, emphasizing communication and trade-offs. Creating an egalitarian partnership is less about the division of labor and more about the couples' relationship related to relative status, attention, accommodation, and well-being (Knudson-Martin & Mahoney, 2009). This study was limited in the degree to which it could respond to a couple's process. While it is interesting to share how a value was transmitted to me, I am only part of the decision-making team in my family. This study could be improved through the use of group interviews or focus groups.

Moreover, autoethnography is a research method that provides an opportunity for people to tell their stories and represent their truths, and can demonstrate how people live, struggle, and achieve (Bochner & Ellis, 2006). It is an intentionally subjective (Ellis & Bochner, 2000) study

of culture and should not be used to make conclusions about entire groups. Criticism of autoethnography as a research method include concerns that personal narratives are not research (Walford, 2004), it is a self-indulgent form of writing (Sparkes, 2002) and that primarily consist of only one data source. However, autoethnography provides a cultural understanding of a phenomena, in this case – egalitarian parenting, and connects perspectives of self to others in context (Ngunjiri, Hernandez, & Chang, 2010). This study does not make claims about the intergenerational transmission of parenting values, rather provides an opportunity to share alternative constructions of family, featuring issues of work-life balance, flexible gender roles, and intersectional experiences.

Suggestions for Further Research

The scholar bell hooks (2000) predicted, "In the future feminist movement, we will spend less time critiquing patriarchal marriage bonds and expend more effort showing alternatives, showing the value of peer relationships which are founded on principles of equality, respect, and the belief that mutual satisfaction and growth are needed for partnerships to be fulfilling and lasting" (p. 84). This prediction has been borne out in more recent studies on egalitarian partnerships (Cowdery et al., 2009; Curenton, Crowley, & Mouzon, 2018; Knudson-Martin & Mahoney, 2009). New egalitarian partnerships may be founded on principles focused on the wellbeing of individuals *and* the partnership, such as respect, mutuality, support, and flexibility. Modern couples struggling with work-family balance may need to focus more on their partnership and less on an equitable division of responsibilities. Today's couples need to focus on effectively sharing responsibilities and distributing leadership to build mutually supportive relationships. Additional research is needed to understand how couples negotiate egalitarian

relationships, work together to achieve the collective good (Cowdery et al., 2009), maintain communication, and strengthen and respect their partnerships and relationships.

Conclusion

Black families, across the life course, should be studied qualitatively using multiple approaches. The stories Black families tell about egalitarianism need to be collected and shared in order to inform a narrative of how Black families share responsibilities. This study was guided by life course theory and informed by the historical context of Black families. This study responded to research questions about how social-historical context, as filtered through previous generations, influenced my conception and practice of egalitarian parenting. Moreover, this inquiry sought to understand how multiple generations (G1, G2, and G3) constructed and practiced egalitarian parenting and how G4 would construct and practice egalitarian parenting. Research methods included critical reflection, journaling, in-depth individual oral history interviews, and the review of photographs and historical documents. Trustworthiness was ensured using repeated data collection, member checks, and the consultation of critical friends.

REFERENCES

- Acock, A. C., & Bengtson, V. L. (1978). On the relative influence of mothers and fathers: A covariance analysis of political and religious socialization. *Journal of Marriage and Family*, 40(3), 519-530. <u>By</u>
- Adams. T. E. & Manning, J. (2015). Autoethnography and family research. *Journal of Family Theory & Review*, 7: 350–366. doi:10.1111/jftr.12116
- Amato, P. R. (1996). Explaining the intergenerational transmission of divorce. *Journal of Marriage and Family*, 58(3), 628-640. http://www.jstor.org/stable/353723
- Ames, B. D. (1999). Lifecourse parenting: The primary intergenerational relationship. *Michigan Family Review*, 4(1), 17-26. http://dx.doi.org/10.3998/mfr.4919087.0004.103
- Andersen, L. H. (2018). Assortative mating and the intergenerational transmission of parental incarceration risks. *Journal of Marriage & Family*, 80(2), 463-477. https://doi-org.proxy.libraries.rutgers.edu/10.1111/jomf.12459
- Andersen, M.L. & Collins, P.H. (2013). Systems of power and inequality. In M.L. Andersen & P.H. Collins (Eds.) *Race, class, and gender: An anthology* (pp. 61-84). Belmont, CA: Wadsworth.
- Angelou, M. (1969). I Know Why the Caged Bird Sings. New York: Random House.
- Armitage, S. & Mercier, L. (2009). Speaking history: Oral histories of the American past, 1865present. New York: Palgrave Macmillan.
- Awosika, M. (2005, December 9). Lottie Swann Crawford [video recording]. The Herald Tribune. Retrieved from https://www.heraldtribune.com/article/LK/20051209/News/605249537/SH/

- Baird, R. (2012). Constructing lives: A new approach to understanding migrants' oral history narratives. *Oral History*, 40(1), 57-66. https://www.jstor.org/stable/41806581
- Banks, J. A. (1998). The lives and values of researchers: Implications for educating citizens in a multicultural society. *Educational Researcher*, *27*(7), 4-17.

 https://www.jstor.org/stable/1176055
- Baumrind, D. (1971). Current patterns of parental authority. *Developmental Psychology*Monograph, 4, 1-103.
- Baumrind, D. (1991). Parenting styles and adolescent development. In J. Brooks-Gunn, R. Lerner, & A. C. Peterson (Eds.), *The encyclopedia of adolescence* (pp.746-758). New York: Garland.
- Becker, P. E., & Moen, P. (1999). Scaling back: Dual-earner couples' work-family strategies. *Journal of Marriage and Family*, 61(4), 995-1007. http://dx.doi.org/10.2307/354019
- Beckett, J. O., & Smith, A. D. (1981). Work and family roles: Egalitarian marriage in Black and white families. *Social Service Review*, 55(2), 314-326.

 http://www.jstor.org/stable/30011475
- Belsky, J., Jafee, S. R., Sligo, J., Woodward, L., & Silva, P. A. (2005). Intergenerational transmission of warm-sensitive-stimulating parenting: A prospective study of mothers and fathers of 3-year-olds. *Child Development*, 76(2), 384-396. http://dx.doi.org/10.1111/j.1467-8624.2005.00852.x
- Bengtson, V. L., & Allen, K. R. (1993). The life course perspective applied to families over time. In P. G. Boss, W. J. Doherty, R. LaRossa, W. R. Schumm, & S. K. Steinmetz (Eds.), Sourcebook of family theories and methods: A contextual approach (pp. 469–504). New York: Plenum.

- Blee, K. M. & Tickamyer, A. R. (1995). Racial differences in men's attitudes about women's gender roles. Journal of Marriage and Family, 57(1), 21-30.

 http://www.jstor.org/stable/353813
- Bochner, A. P., & Ellis, C. (2006). Communication as autoethnography. In G. J. Shepherd J. S. John & T. Striphas (Eds.), *Communication as ...: Perspectives on theory* (pp. 13-21). Thousand Oaks, CA: SAGE Publications, Inc. doi: 10.4135/9781483329055.n13
- Borland, K. (1998). 'That's Not What I Said': Interpretative Conflict in Oral Narrative Research.

 In R. Perks & A. Thomson (Eds.), *The oral history reader* (pp. 310-321). New York:

 Routledge.
- Boylorn, R.M., & Orbe, M.P. (2014). Critical Autoethnography: Intersecting Cultural Identities in Everyday Life. New York, NY: Routledge.
- Brown, G. W. (1997). Oral History: Louisiana Black Women'S Memories. In K.M. Vaz (Ed.),

 Oral narrative research with black women (pp.83-95). Thousand Oaks, CA: SAGE

 Publications, Inc.
- Brown, L.M. & Gilligan, C. (1991). Listening for voice in narratives of relationship. *New Directions for Child Development*, *54*, 43-62.
- Brunsma, D. L. (2006). Public categories, private identities: Exploring regional differences in the biracial experience. *Social Science Research*, *35*, 555–576.

 doi:10.1016/j.ssresearch.2004.10.002
- Bulanda, R. E. (2004). Paternal involvement with children: The influence of gender ideologies. *Journal of Marriage and Family*, 66, 40-45.

- Burton, L., Bonilla-Silva, E., Ray, V., Buckelew, R., & Freeman, E. (2010). Critical Race theories, colorism, and the decade's research on families of color. *Journal of Marriage and Family*, 72(3), 440-459. www.jstor.org/stable/40732490
- Bus, A. G., van IJzendoorn, M. H., & Pellegrini, A. D. (1995). Joint book reading makes for success in learning to read: A meta-analysis on intergenerational transmission of literacy.
 Review of Educational Research, 65(1), 1-21.
 https://doi.org/10.3102/00346543065001001
- Carlo, G., White, R. M. B., Streit, C., Knight, G. P., & Zeiders, K. H. (2018). Longitudinal relations among parenting styles, prosocial behaviors, and academic outcomes in U.S. Mexican adolescents. *Child Development*, 89(2), 577-592. doi: 10.1111/cdev.12761
- Carlson, M. J., VanOrman, A. G., & Pilkauskas, N. V. (2013). Examining the antecedents of U.S. nonmarital fatherhood. *Demography*, *50*, 1421–1447. http://dx.doi.org/10.1007/s13524-013-0201-9
- Carr, D. S. (2004). "My daughter has a career—I just raised babies": Women's intergenerational social comparisons. *Social Psychology Quarterly*, *67*, 132–154. http://www.jstor.org/stable/3649083
- Chang, H. (2008). Autoethnography. In H. Chang (Ed.) *Autoethnography as method*, (pp. 43-57). Walnut Creek, CA: Left Coast Press.
- Chancler, L. M., Webb, F. J. & Miller, C. (2017). Role of the Black grandmother in the racial socialization of their biracial grandchildren. *Marriage & Family Review*, *53*(5), 444-464, DOI: 10.1080/01494929.2017.1297755

- Chavez, C. (2008). Conceptualizing from the inside: Advantages, complications, and demands on insider positionality. *The Qualitative Report*, *13*(3), 474-494. Retrieved from https://nsuworks.nova.edu/tqr/vol13/iss3/9
- Chen, Z., & Kaplan, H. B. (2001), Intergenerational transmission of constructive parenting.

 *Journal of Marriage and Family, 63: 17–31. http://dx.doi.org/10.1111/j.1741-3737.2001.00017.x
- Chao, R. K. (2001) Extending research on the consequences of parenting style for Chinese Americans and European Americans. *Child Development*, 72, 1832-1843.

 DOI:10.1111/1467-8624.00381
- Cohany, S. R., & Sok, E. (2007). Trends in labor force participation of married mothers of infants. *Monthly Labor Review*. http://www.bls.gov/opub/mlr/2007/02/art2full.pdf
- Connor, M. E., & White, J. L. (2011). *Black fathers: An invisible presence in America*. (Second edition). New York, NY: Routledge.
- Cowdery, R. S., Scarborough, N., Knudson-Martin, C., Seshadri, G., Lewis, M. E., & Mahoney, A. R. (2009). Gendered power in cultural contexts: Part II. Middle class African American heterosexual couples with young children. *Family Process*, 48, 25–39. doi:10.1111/j.1545-5300.2009.01265.x
- Clandinin, D.J. & Connelly, F.M. (2000). *Narrative inquiry: Experiences and story in qualitative research*. San Francisco: Jossey-Bass.
- Cleveland, H. (1886). Alexander H. Stephens, in Public and Private: With Letters and Speeches, Before, During, and Since the War. Philadelphia, 717-729.

- Cochran, D. (1997). African American fathers: A decade review of the literature. Families in Society: The Journal of Contemporary Social Services, 78(4), 340-351. https://doi.org/10.1606/1044-3894.792
- Connelly, F.M. & Clandinin, D.J. (1990). Stories of experience and narrative inquiry. *Educational Researcher*, 19(5), 2-14. doi: 10.3102/0013189X019005002
- Coontz, S. (2011). A Strange Stirring: The Feminine Mystique and American Women at the Dawn of the 1960s. Philadelphia, PA: Basic Books.
- Cotter, D., Hermsen, J. M., & Vanneman, R. (2011). The end of the gender revolution? Gender role attitudes from 1977 to 2008. *American Journal of Sociology*, 117(1), 259–289. http://www.jstor.org/stable/10.1086/658853
- Crawford, L. K. S. (2014, February 18). Interview by Brenda Ross [audio recording] Part One.

 **Library Association of Rockland County*. New City Library Oral History Collection, New City, NY. Retrieved from https://cdm16694.contentdm.oclc.org/digital/collection/larc/id/3212/rec/1
- Crawford, L. K. S. (2017, February 18). Interview by Brenda Ross [audio recording] Part Two.

 **Library Association of Rockland County*. New City Library Oral History Collection, New City, NY. Retrieved from https://cdm16694.contentdm.oclc.org/digital/collection/larc/id/3213
- Crawford, L. K. S. (2017, February 18). Interview by Brenda Ross [audio recording] Part Thre e. *Library Association of Rockland County*. New City Library Oral History Collection, N ew City, NY. Retrieved from https://cdm16694.contentdm.oclc.org/digital/collection/larc/id/3214

- Crenshaw, K. (1989). Demarginalizing the intersection of race and sex: A black feminist critique of antidiscrimination doctrine, feminist theory and antiracist politics. *University of Chicago Legal Forum*, 1989(1). http://chicagounbound.uchicago.edu/uclf/vol1989/iss1/8
- Csizmadia, A., Brunsma, D. L. & Cooney, T. M. (2012). Racial identification and developmental outcomes among Black—White multiracial youth: A review from a life course perspective. *Advances in Life Course Research*, *17*, 34-44. doi:10.1016/j.alcr.2011.10.003
- Cunningham, M. (2001). The influence of parental attitudes and behaviors on children's attitudes toward gender and household labor in early adulthood. *Journal of Marriage and Family*, 63, 111–122. http://dx.doi.org/10.1111/j.1741-3737.2001.00111.x
- Curenton, S.M., Crowley, J.E. & Mouzon, D.M. (2018). Qualitative descriptions of middle-class, African American mothers' child rearing practices and values. *Journal of Family Issues*, 39(4), 868-895. http://dx.doi.org/doi:10.7282/T39026ZP
- Daly, K. J. (1996). Families & Time: Keeping Pace in a Hurried Culture. Thousand Oaks, CA: SAGE Publications, Inc. http://dx.doi.org/10.4135/9781483327792
- Daly, K. J. (2001). Deconstructing family time: From ideology to lived experience. *Journal of Marriage and Family*, 63(2), 283-294. http://www.jstor.org/stable/3654592
- Daly, K. J. (2002). Time, gender, and the negotiation of family schedules. *Symbolic Interactionism*, 25(3), 323-342. https://doi.org/10.1525/si.2002.25.3.323
- Daly, K. J. (2003). Family theory versus the theories families live by. *Journal of Marriage and Family*, 65(4), 771-784. http://www.jstor.org/stable/3599889
- Daly, K. J. (2007). *Qualitative Methods for Family Studies & Human Development*. Thousand Oaks, CA: SAGE Publications, Inc.

- Dean, P., Marsh, K., Landry, B. (2015). Cultural Contradiction or Integration? Work–Family

 Schemas of Black Middle Class Mothers. *Notions of Family: Intersectional Perspectives*,

 137-158. https://doi.org/10.1108/S1529-2126(2013)0000017010
- Denzin, N. K. & Lincoln, Y. S. (2000). Handbook of Qualitative Research, 2nd edition.

 Thousand Oaks, CA: SAGE Publications, Inc.
- Dietrich, K.T. (1975). A reexamination of the myth of Black matriarchy. *Journal of Marriage* the Family, 37(2), 367-374. http://www.jstor.org/stable/350970
- Dill, B. T. (1988). Our mothers' grief: Racial ethnic women and the maintenance of families. In
 M.L. Anderson and P.H. Collins (eds.) *Race, Class, and Gender: An Anthology* (pp. 314-326). Belmont, CA: Wadsworth.
- Dornbusch, S. M., Ritter, P. L., Leiderman, P. H., Roberts, D. F., & Fraleigh, M. J. (1987). The relation of parenting style to adolescent school performance. *Child Development*, *58*, 1244-1257.
- Du Bois, W. E. B. (1903). The souls of Black folk. Middletown: Millennium Publications.
- Dugger, K. (1988). Social location and gender-role attitudes: A comparison of Black and white women. *Gender and Society*, *2*(4), 425-448. http://www.jstor.org/stable/190208
- Dyck, V. & Daly, K.J. (2006). Rising to the challenge: Fathers' role in the negotiation of couple time. *Leisure Studies*, 25(2), 201-217. doi: 10.1080/02614360500418589
- Ehrensaft, M. K., Cohen, P., Brown, J., Smailes, E., Chen, H., & Johnson, J. G. (2003).

 Intergenerational transmission of partner violence: A 20-year prospective study. *Journal of Consulting and Clinical Psychology*, 71,741-753.

 http://204.14.132.173/pubs/journals/releases/ccp-714741.pdf

- Elder, G. H., Jr. (1974/1999). *Children of the Great Depression: Social change in life*experience. Chicago: University of Chicago Press. (Original work published 1974)
- Elder, G. H. (1998). The life course as developmental theory. *Child Development*, 69 (1), 1-12. http://dx.doi.org/10.2307/1132065
- Elder, G. H., Shanahan, M. J., & Jennings, J. A. (2015). Human development in time and place.
 In R. M. Lerner, M. Bornstein, & T. Leventhal (Eds.), *Handbook of child psychology and developmental science* (7th ed., Vol. 4, pp. 6–54). Hoboken, NJ: John Wiley & Sons.
 https://doi.org/10.1002/9781118963418.childpsy402
- Elder, G. H., Jr., Wu, W., & Yuan, J. (1993). State-initiated change and the life course in Shanghai, China. Unpublished project report.
- Ellis, C., & Bochner, A. P. (2000). Autoethnography, personal narrative, reflexivity: Researcher as subject. In N. K. Denzin & Y. S. Lincoln (Eds.), *Handbook of Qualitative Research* (pp. 733-768). London: Sage.
- Ellis, C. & Bochner, A.P. (2005). Autoethnography, personal narrative, reflexivity: Researcher as subject. In N.K. Denzin & Y.S. Lincoln (Eds.), *Collecting and interpreting qualitative materials Second edition* (pp. 199-258). Thousand Oaks, CA: SAGE Publications, Inc. https://www.researchgate.net/publication/276935293
- Few, A. (2007). Integrating Black consciousness and critical face feminism into family studies research. *Journal of Family Issues*, 28, 452-473.
- Fletcher, A. C., Steinberg, L., & Sellers, E. B. (1999). Adolescents' well-being as a function of perceived interparental inconsistency. *Journal of Marriage and Family, 61*, 599-610.
- Ford, D. H., & Lerner, R. M. (1992). *Developmental systems theory: An integrative approach*.

 Thousand Oaks, CA: SAGE Publications, Inc.

- Frazier, E.F. (1939). *The Negro family in the United States*. Chicago: The University of Chicago Press.
- Frazier, E.F. (1948). Ethnic family patterns: The Negro family in the United States American. *Journal of Sociology*, 53(6), 435-438.
- Furstenberg, F. (2007). The making of the Black family: Race and class in qualitative studies in the twentieth century. *Annual Review of Sociology, 33*, 429-448. doi: 10.1146/annurev.soc.33.040406.131727.
- Gadsden, V.L., & Smith, R.R. (1994). African American males and fatherhood: Issues in research and practice. *Journal of Negro Education*, *63*(4), 634-648. Retrieved from http://www.jstor.org/stable/2967300.
- Garcia, F. & Garcia, E. (2009). Is authoritative always the optimum parenting style? Evidence from Spanish families. *Adolescence*, *44*(173), 101-131. http://pascal-francis.inist.fr/vibad/index.php?action=search&terms=21330391
- Gaunt, R. & Bassi, L. (2012). Modeling and compensatory processes underlying involvement in child care among kibbutz-reared fathers. *Journal of Family Issues*, *33*(6), 823-848. doi: 10.1177/0192513X11428440
- Gerson, K. (2010). The unfinished revolution: Coming of age in a new era of gender, work, and family. New York, NY: Oxford University Press.
- Gilligan, C. (2015). The listening guide method of psychological inquiry. *Qualitative Psychology*, 2, 69–77. http://dx.doi.org/10.1037/qup0000023
- Gilligan, C., Spencer, R., Weinberg, M. K., & Bertsch, T. (2006). On the listening guide: A voice-centered relational method. In S. N. Hesse-Biber & P. Leavy (Eds.), *Emergent methods in social research* (pp. 253–271). Thousand Oaks, CA: SAGE Publications, Inc.

- Goldin, C. (1977). Female labor force participation: The origin of Black and white differences, 1870 and 1880. *Journal of Economic History, 37*(1), 87-108. http://www.jstor.org/stable/2119448
- Glass, J., Bengtson, V. L., & Dunham, C. C. (1986). Attitude similarity in three generation families: Socialization, status inheritance, or reciprocal influence? *American Sociological Review*, *51*, 685–698. http://www.jstor.org/stable/2095493
- Glauber, R., & Gozjolko, K. L. (2011). Do traditional fathers always work more? Gender ideology, race, and parenthood. *Journal of Marriage and Family, 73*, 1133-1148. http://dx.doi.org/10.1111/j.1741-3737.2011.00870.x
- Golding, B. & Hargreaves, J. (2018). Humanitarian nursing with Médecins Sans Frontières:

 Foregrounding the listening guide as a method for analysing oral history data. *Journal of Advanced Nursing*, 74(8), 1984-1992. doi: 10.1111/jan.13585
- Greenfield, E., & Marks, N. (2006). Linked lives: Adult children's problems and their parents' psychological and relational well-being. *Journal of Marriage and Family*, 68, 442–454. https://www.ncbi.nlm.nih.gov/pmc/articles/PMC1950122/
- Gutman, H.G. (1976). *The Black family in slavery and freedom, 1750-1925*. New York: First Vintage Books.
- Guy, M. E., & Fenley, V. M. (2014). Inch by inch: Gender equity since the Civil Rights Act of 1964. *Review of Public Personnel Administration*, 34(1), 40–58. http://dx.doi.org/10.1177/0734371X13510379
- Habib, C. (2012). The transition to fatherhood: A literature review exploring paternal involvement with identity theory. *Journal of Family Studies*, *18*(2-3), 103-120. doi: 10.5172/jfs.2012.2606

- Hamer, J. (1997). The fathers of "fatherless" Black children. *Families in Society*, 78, 564-578. https://doi.org/10.1606/1044-3894.3387
- Hareven, T. (1994). Aging and generational relations: A historical and life course perspective. *Annual Review of Sociology*, 20, 437-461.
- Hayano, D. M. (1979). Auto-ethnography: Paradigms, problems, and prospects. *Human Organization*, 38(1), 99-104. doi:10.17730/humo.38.1.u761n5601t4g318v
- Heckert, D. A., Nowak, T. C., & Snyder, K. A. (1998). The impact of husbands' and wives' relative earnings on marital distribution. *Journal of Marriage and the Family*, 60, 690–703. doi:10.1108/2FS1529-2126/282013/290000017014
- Hofner, C., Schadler, C., & Richter, R. (2011). When men become fathers: Men's identity at the transition to parenthood. *Journal of Comparative Family Studies*, 42(5), 667-686. https://www.jstor.org/stable/41604478
- hooks, b. (2004). Culture to culture: Ethnography and cultural studies as critical intervention. In In S. N. Hesse-Biber & P. Leavy (Eds.), *Approaches to qualitative research: A reader on theory and practice*, New York, NY: Oxford University Press, 149-158.
- House, G. (2010). We'll Never Turn Back. In Holsaert, Noonan, Richardson, Robinson, Young & Zellner. (Eds.), *Hands on the freedom plow: Personal accounts by women in SNCC* (pp. 503-514). Urbana, Chicago, and Springfield: University of Illinois Press. www.jstor.org/stable/10.5406/j.ctt1hj9xfc.64
- Hughes, L. (1926). The Weary Blues. New York: A. A. Knopf.
- Huguley, J. P., Wang, M., Vasquez, A. C. & Guo, J. (2019). Parental ethnic-racial socialization practices and the construction of children of color's ethnic-racial identity: A research

- synthesis and meta-analysis. *Psychological Bulletin, 145*(5), 437-458. http://dx.doi.org/10.1037/bul0000187
- Humphreys, M. (2005). Getting personal: Reflexivity and autoethnographic vignettes.

 *Qualitative Inquiry, 11(6), 840-860.
- Humphreys, M., Coupland, C. & Learmouth, M. (2015). Autoethnographic vignettes of working lives. Presented at the Asia Pacific Researchers in Organization Studies (APROS) and European Group for Organizational Studies (EGOS) Conference, Sydney, Australia, Dec 9-11. https://dspace.lboro.ac.uk/2134/20243
- Jarrett, R. L. (2000). Neighborhood effects models: A view from the neighborhood. *Research in Community Sociology*, 10, 305-323.
- Kerr, M., Stattin, H., Biesecker, G., & Ferrer-Wreder, L. (2003). Relationships with parents and peers in adolescence. In R. M. Lerner, M. A. Easterbrooks, & J. Mistry (Eds.), *Handbook* of psychology. Vol. 6. Developmental psychology. Hoboken, NJ: Wiley, pp. 395-419. DOI:10.1002/0471264385.wei0616.
- Johnson, G. B. (1954). A sociologist looks a racial desegregation in the South. *Social Forces*, 33, 1-10.
- Jones, J., & Mosher, W. D. (2013, December 20). Fathers' involvement with their children: United States, 2006–2010. *National Health Statistics Reports, 71*. Hyattsville, MD: National Center for Health Statistics. https://www.cdc.gov/nchs/data/nhsr/nhsr07
- Katovich, M. A. & Couch, C.J. (1992). The nature of social pasts and their use as foundations for situated action. *Symbolic Interaction*, 15(1), 25-47.
 - http://www.jstor.org/stable/10.1525/si.1992.15.1.25

- Kerr, D. C. R., Capaldi, D. M., Pears, K. C., & Owen, L. D. (2009). A prospective three generational study of fathers' constructive parenting: Influences from family of origin, adolescent adjustment, and offspring temperament. *Developmental Psychology*, 45(5), 1257–1275. https://doi.org/10.1037/a0015863
- Kertzer, D. I. (1983). Generation as a sociological problem. *Annual Review of Sociology, 9*, 125-149. http://www.jstor.org/stable/2946060
- Khanna, N. (2010). If you're half Black, you're just Black'': Reflected appraisals and the persistence of the one-drop rule. *The Sociological Quarterly*, *51*, 96–121. https://www.jstor.org/stable/20697932
- Knudson-Martin C., Mahoney A. R. (2009). Introduction to the special section-Gendered power in cultural contexts: Capturing the lived experience of couples. *Family Process*, 48, 5-8. https://doi.org/10.1111/j.1545-5300.2009.01263.x
- Lincoln, Y.S. & Guba, E.G. (1985). *Naturalistic Inquiry*. Newbury Park, CA: SAGE Publications, Inc.
- Luttrell, W. (2005). 'Good enough' methods for life-story analysis. In N. Quinn, *Finding culture* in talk: A collection of methods (pp. 243-268). United States: Palgrave Macmillan. doi: 10.1007/978-1-137-05871-3 7.
- Luttrell, W. (2010a). Interactive and reflexive models of qualitative research design. In W. Luttrell (Ed.), *Qualitative educational research: Readings in reflective methodology and transformative practice* (pp. 159-163). London: Routledge.
- Luttrell, W. (2010b). Reflexive writing exercises. In W. Luttrell (Ed.), *Qualitative educational* research: Readings in reflective methodology and transformative practice (pp. 469–480). London: Routledge.

- Kane, E. W. (2000). Racial and ethnic variations in gender-related attitudes. *Annual Review of Sociology*, 26, 419-439. http://dx.doi.org/10.1146/annurev.soc.26.1.419
- Ko, H. C., Wang, L. L., & Xu, Y. T. (2013). Understanding the different types of social support offered by audience to A-list diary-like and informative bloggers. *Cyberpsychology*, behavior and social networking, 16(3), 194-199. https://doi.org/10.1089/cyber.2012.0297
- Mack, D. (1978). The power relationship in Black families and white families. In R. Staples (ed.), *The Black family: Essays and studies*. Belmont, CA: Wadsworth.
- Macmillan, R., & Copher, R. (2005). Families in the life course: Interdependency of roles, role configurations, and pathways. *Journal of Marriage and Family*, *67*, 858–879. https://doi.org/10.1111/j.1741-3737.2005.00180.x
- Majumder, M. A. (2016). The impact of parenting style on children's educational outcomes in the United States. *Journal of Family Economic Issues*, *37*, 89–98. https://doi.org/10.1007/s10834-015-9444-5
- Masterson, C. R. & Hoobler, J.M. (2015). Care and career: A family identity-based typology of dual-earner couples. *Journal of Organizational Behavior*, *36*, 75-93. DOI: 10.1002/job.1945
- Maxwell, J. A. (2013). *Qualitative Research Design, an Interactive Approach* (Third Ed.). Thousand Oaks, CA: Sage Publications.
- Mbiti, J. (1969/1997). African religions & philosophy. Oxford: Heinemann.
- McAdoo, H.P. & McAdoo, J.L. (1997). The dynamics of African American fathers' family roles. *Michigan Family Review*, *3*(1), 7-15. Retrieved from http://hdl.handle.net/2027/spo.4919087.0003.102

- McGill, B. S. (2014). Navigating new norms of involved fatherhood: Employment, fathering attitudes, and father involvement. *Journal of Family Studies*, *35*(8), 1089–1106. https://doi.org/10.1177/0192251X14522247
- McKay, R. B. (2007). A Black community with advanced labor force characteristics in 1960.

 *Monthly Labor Review, 130(2), 3-8. https://www-jstor-org.proxy.libraries.rutgers.edu/stable/23806132
- McKee, J.B. (1993). Sociology and the race problem: The failure of a perspective. Board of Trustees at the University of Illinois. Urbana and Chicago, IL.
- Messner, M. A. (1993). "Changing men" and feminist politics in the United States. *Theory and Society*, 22(5), 723-737. http://www.jstor.org/stable/657993
- Middleton, R. & Putney, S. (1960). Dominance in decisions in the family: Race and class differences. In C.V. Willie (ed.), *The Family Life of Black People*. Columbus, OH: Charles E. Merrill.
- Miller, W. L. & Crabtree, B. F. (2004). Depth interviewing. In S. N. Hesse-Biber & P. Leavy (Eds.), *Approaches to Qualitative Research: A reader on theory and practice* (pp. 185-202). New York, NY: Oxford University Press
- Min, J., Silverstein, M., & Lendon, J.P. (2012). Intergenerational transmission of values over the family life course. Advances in Life Course Research, 17, 112–120.
 http://dx.doi.org/10.1016/j.alcr.2012.05.001
- Mizzi, R. (2010). Unraveling researcher subjectivity through multivocality in autoethnography.

 Journal of Research Practice, 6(1).

 http://jrp.icaap.org/index.php/jrp/article/view/201/185

- Moen, P., Erickson, M. A., & Dempster-McClain, D. (1997). Their mother's daughters? The intergenerational transmission of gender attitudes in a world of changing roles. *Journal of Marriage and the Family, 59*, 281-293. http://www.jstor.org/stable/353470
- Moody, A. (1968). Coming of age in Mississippi. New York, NY: Bantam Dell.
- Mowen, T. J. & Schroeder, R. D. (2018). Maternal parenting style and delinquency by race and the moderating effect of structural disadvantage. *Youth & Society*, 50(2), 139-159. doi: 10.1177/0044118X15598028
- Moynihan, D.P. (1965). The Negro family in America: The case for national action. Washington,
 D.C.: Office of Policy Planning and Research, U.S. Department of Labor.

 https://www.dol.gov/oasam/programs/history/webid-meynihan.htm
- Nash, R. J. (2004). *Liberating scholarly writing: The power of personal narrative*. New York, NY: Teachers College Press.
- Neblett, E. W., Jr., Smalls, C. P., Ford, K. R., Nguyên, H. X., & Sellers, R. M. (2009). Racial socialization and racial identity: African American parents' messages about race as precursors to identity. *Journal of Youth and Adolescence*, 38, 189-203. http://dx.doi.org/10.1007/s10964-008-9359-7
- Ngunjiri, F. W., Hernandez, K. C., & Chang, H. (2010). Living autoethnography: Connecting life and research [Editorial]. *Journal of Research Practice*, 6(1), Article E1. http://jrp.icaap.org/index.php/jrp/article/view/241/186
- Nussbaum, B. (2003). African culture and Ubuntu: Reflections of a South African in America.

 *Perspectives, 17(1), 1-12.
 - https://worldbusiness.org/wpcontent/uploads/2013/05/pr050703.pdf
- Ochs, E. & Capps, L. (1996). Narrating the self. Annual Review of Anthropology, 25, 19-43 -

- Orbuch, T. L., & Eyster, S. L. (1997). Division of household labor among black couples and white couples. *Social Forces*, 76, 301-332. http://dx.doi.org/10.2307/2580327
- Overton, W. F. (2015). Processes, relations and Relational-Developmental-Systems. In Overton & Molenaar (Eds.), *Theory and method: Volume 1 of the handbook of child psychology and developmental science Seventh edition* (pp. 9-62). Hoboken, NJ: Wiley, 9-62.
- Peaks, J. S. (2014, September 21). Interview by Eileen McAdam [audio recording]. Part One Sound and Story Project. *Library Association of Rockland County*. New City Library Oral History Collection, New City, NY. Retrieved from
- Peaks, J. S. (2014, September 21). Interview by Eileen McAdam [audio recording] Part Two. S ound and Story Project. *Library Association of Rockland County*. New City Library Oral History Collection, New City, NY. Retrieved from https://cdm16694.contentdm.oclc.org/digital/collection/larc/id/3083
- Perry, A. R., Harmon, D. K., & Leeper, J. (2012). Resident Black fathers' involvement: A comparative analysis of married and unwed cohabitating fathers. *Journal of Family Issues*, *33*(6), 695-714. http://dx.doi.org/10.1177/0192513X11428125
- Pettigrew, T. (1980). *The sociology of race relations: Reflection and reform*. New York: The Free Press.
- Plessy v. Ferguson, 163 U.S. 537, 1896
- Platt, L., & Polavieja, J. (2016). Saying and doing gender: Intergenerational transmission of attitudes towards the sexual division of labour. *European Sociological Review, 32*(6), 820–834. http://dx.doi.org/10.1093/esr/jcw037

- Perry, A.R., Harmon, D.K., & Leeper, J. (2012). Resident Black fathers' involvement: A comparative analysis of married and unwed cohabitating fathers. *Journal of Family Issues*, *33*(6), 695, DOI: -10.1177/0192513X11428125.
- Philaretou, A. G. & Allen, K.R. (2006) (2006). Researching sensitive topics through autoethnographic means. *The Journal of Men's Studies*, *14*(1), 65-78. https://doi.org/10.3149/jms.1401.65
- Portelli, A. (1998). What Makes Oral History Different? In R. Perks & A. Thomson. (Eds.), *The oral history reader* (pp. 63-74). New York: Routledge. 63-74.
- Raley, S. B, Mattingly, M. J., & Bianchi, S. M. (2006). How dual are dual-income couples?

 Documenting change from 1970 to 2001. *Journal of Marriage and Family, 68*, 11-28. https://doi.org/10.1111/j.1741-3737.2006.00230.x
- Ray, R. & Jackson, P. B., (2015). Intersectionality and work–family balance: A study of Black, white, and Mexican-American adults. In M.H. Kohlman, D. B. Krieg & B. J. Dickerson (Eds.), *Notions of family: Intersectional perspectives, advances in gender research*, (Vol. 17, pp. 241-262). Emerald Group Publishing Limited. https://doi.org/10.1108/S1529-2126(2013)0000017014
- Real, D. (1989). Super Media: A Cultural Studies Approach. Newbury Park, CA: SAGE Publications, Inc.
- Richie, D. A. (2003) *Doing oral history Second Edition*. New York: Oxford University Press. https://epdf.tips/doing-oral-history-a-practical-guide.html
- Rivas-Drake, D., Hughes, D., & Way, N. (2009). A preliminary analysis of associations among ethnic–racial socialization, ethnic discrimination, and ethnic identity among urban sixth

- graders. *Journal of Research on Adolescence*, *19*, 558–584. http://dx.doi.org/10.1111/j.1532-7795.2009.00607.x
- Roest, A. M. C., Dubas, J.S., & Gerris, J.R.M. (2010). Value transmissions between parents and children: Gender and developmental phase as transmission belts. *Journal of Adolescence*, 33, 21–31. doi:10.1016/j.adolescence.2009.05.017
- Ryu, S., & Mortimer, J. T. (1996). The "occupational linkage hypothesis" applied to occupational value formation in adolescence. In J. T. Mortimer & M. D. Finch (Eds.), *Adolescents, work, and family: An intergenerational developmental analysis* (Vol. 6, pp. 167–190). Thousand Oaks, CA: SAGE Publications, Inc.
- Sabattini, L., & Leaper, C. (2004). The relation between mother's and father's parenting styles and their division of labor in the home: Young adults' retrospective reports. *Sex Roles*, 50, 217-225.
- Sandelowski, M. (1991). Telling stories: Narrative approaches in qualitative research. *Journal of Nursing Scholarship*, 23(3), 161-166. doi: 10.1111/j.1547-5069.1991.tb00662.x.
- Sanders Thompson, V. L. (1994). Socialization to race and its relationship to racial identification among African Americans. *The Journal of Black Psychology*, 20(2), 175–188. doi:10.1177/00957984940202006.
- Scanzoni, J. H. (1975). Sex roles, economic factors, and marital stability in Black and white marriages. *Journal of Marriage and the Family, 37*, 130-144.
- Schofield, T. J., & Abraham, W. T. (2017). Intergenerational continuity in attitudes: A latent variable family fixed-effects approach. *Journal of Family Psychology*, 31(8), 1005–1016. http://dx.doi.org/10.1037/fam0000375

- Scott, D. v. Sanford, J. F.A. (1857, March 6). Judgment in the U.S. Supreme Court Case *Dred Scott* v. *John F.A. Sanford*, 1857; Case Files 1792-1995; Record Group 267; Records of the Supreme Court of the United States; National Archives.
 https://www.ourdocuments.gov/doc.php?flash=true&doc=29
- Seibold, C. (2001). Qualitative research from a feminist perspective in the postmodern era: Methodological, ethical and reflexive concerns. *Nursing Inquiry*, 7, 147-155. 10.1046/j.1440-1800.2000.00063.x.
- Simons, L. G. & Conger, R. D. (2007). Linking mother–father differences in parenting to a typology of family parenting styles and adolescent outcomes. *Journal of Family Issues*, 28(2), 212-241. doi: 10.1177/0192513X06294593
- Sorkhabi, N. (2012). Parent socialization effects in different cultures: Significance of directive parenting. Psychological Reports, 110(3), 854-878.

 doi: 10.2466/10.02.17.21.PR0.110.3.854-878
- Sosulski, M. R., Buchanan, N. T., & Donnell, C. M. (2010). Life history and narrative analysis: Feminist methodologies contextualizing Black women's experiences with severe mental illness. *Journal of Sociology & Social Welfare, 37*(3), 29-57. Retrieved from http://scholarworks.wmich.edu/cgi/viewcontent.cgi?article=3534&context=jssw
- Sparkes, A. C. (2002). Autoethnography: Self-indulgence or something more? In Bochner, A., Ellis, C. (Eds.), Ethnographically speaking: Autoethnography, literature, and aesthetics. New York: AltaMira.
- Steinberg, L., Elmen, J. D., & Mounts, N. (1989). Authoritative parenting, psychosocial maturity, and academic success in adolescents. *Child Development*, 60, 1424-1436.

- Steinberg, L., Lamborn, S., Dornbusch, S., & Darling, N. (1994). Over-time changes in adjustment and competence among adolescents from authoritative, authoritarian, indulgent, and neglectful families. *Child Development*, 65, 754-770.
- Stephens, A. H. (1861, March 21). Cornerstone Speech: Savannah, Georgia.

 http://teachingamericanhistory.org/library/document/cornerstone-speech/
- Swanson, D.M. (2007). Ubuntu: An African contribution to (re)search for/with a 'humble togetherness.' *Journal of Contemporary Issues in Education*, 2(2), 53-67, http://ejournals.library.ualberta.ca/index.php/JCIE
- Taylor, R. J., Chatters, L. M, & Jackson, J. S. (1993). A profile of familial relations among three-generation Black families. *Family Relations*, 42(3), 332-341.
 http://www.jstor.org/stable/585563
- TenHouten, W. D. (1970). The Black family: Myth and reality. Psychiatry, 23, 145-173.
- Thomas, G., Farrell, M. P., & Barnes, G. M. (1996). The effects of single-mother families and nonresident fathers on delinquency and substance abuse in Black and White adolescents.

 *Journal of Marriage and the Family, 58, 884–894.
- Thornton, A., Alwin, D. F. & Camburn, D. (1983). Causes and consequences of sex-role attitudes and attitude change. *American Sociological Review*, 48(2), 211-227. http://www.jstor.org/stable/2095106
- Thornton, A., & Young-DeMarco, L. (2001). Four decades of trends in attitudes toward family issues in the United States: The 1960s through the 1990s. *Journal of Marriage and Family*, 63, 1009-1037. http://www.jstor.org/stable/3599811

- Trask, B. S. (2017). Integrating life course, globalization and the study of racial and ethnic families. *Journal of Family Theory & Review, 10*(2), 451-466.

 https://doi.org/10.1111/jftr.12259
- Turner, D.D. (1997). Reconstructing the history of musician's protective union local 274 through oral narrative method. In K.M. Vaz (Ed.), *Oral narrative research with black women* (pp.177-196). Thousand Oaks, CA: SAGE Publications, Inc.
- Umberson, D., Pudrovska, T., & Reczek, C. (2010). Parenthood, childlessness, and well-being:

 A life course perspective. *Journal of Marriage and Family*, 72(3), 612–629.

 http://doi.org/10.1111/j.1741-3737.2010.00721.x
- U.S. Constitution amendments. XIII, XIV, XV
- van Eeden-Moorefield, B. & Shih, K. (2015). Family diversity. *The Blackwell Encyclopedia of Sociology*, 4, 1774-1776. https://doi.org/10.1002/9781405165518.wbeosf012.pub2
- Vaz, K.M. (1997). *Oral narrative research with black women*. Thousand Oaks, CA: SAGE Publications, Inc.
- Walford, G. (2004). Finding the limits: Autoethnography and being an Oxford University proctor. *Qualitative Research*, 4(3), 403-417. doi: 10.1177/1468794104047238
- Wall, S. S. (2016). Toward a moderate autoethnography. *International Journal of Qualitative Methods*, 1-9. doi: 10.1177/1609406916674966
- Weis, R. & Toolis, E. (2010). Parenting across cultural contexts in the USA: Assessing parenting behavior in an ethnically and socioeconomically diverse sample. *Early Child Development and Care, 180*, 849-867. doi: 10.1080/03004430802472083.

- Western, B. & Wildemen, C. (2009). The Black family and mass incarceration. *The Annals of the American Academy of Political and Social Sciences*, 62, 221-242. doi:10.1177/0002716208324850
- White, A. M. (2006a). African-American feminist fathers' narratives of parenting. *Journal of Black Psychology*, 32(1), 43-71. http://dx.doi.org/10.1177/0095798405283528
- White, A. M. (2006b). African-American feminist masculinities: personal narratives of redemption, contamination, and peak turning points. *Journal of Humanistic Psychology*, 46(3), 255-280. http://dx.doi.org/10.1177/0022167806286262
- White, D. G. (1999). *Too heavy a load: Black women in defense of themselves 1894-1994*. New York, NY: W.W. Norton & Company.
- Wildeman, C., & Wakefield, S. (2014). The long arm of the law: The concentration of incarceration in families in the era of mass incarceration. *The Journal of Gender, Race & Justice, 17*(2), 367–389.

 http://www.heinonline.org/HOL/Page?handle=hein.journals/jgrj17&div=21
- Wilkerson, I. (2010). The warmth of other suns: The epic story of America's great migration.

 New York City: Vintage Books.
- Williams, R.Y. (2001). I'm a keeper of information: History-telling and voice. *Oral History Review*, 28, 41.
- Williams, R.Y. (2004). The politics of public housing: Black women's struggles against urban inequality. New York: Oxford University Press.
- Williamson, S.H. & Cain, L.P. (2016). Measuring slavery in 2011 dollars. *Measuring Worth*.

 Retrieved from: https://www.measuringworth.com/slavery.php
- Willie, C. V. (1976). A New Look at Black Families. Bayside, NY: General Hall.

- Willie, C. V., & Greenblatt, S. L. (1978). Four 'classic' studies of power relationships in Black families: A review and look to the future. *Journal of Marriage and the Family*, 40(4), 691-694. http://www.jstor.org/stable/351189
- Woodcock, C. (2016). The Listening Guide: A how-to approach on ways to promote educational democracy. *International Journal of Qualitative Methods*, 15(1), 1–10. https://doi.org/10.1177/1609406916677594
- Woodson, J. (2014). Brown Girl Dreaming. New York: Nancy Paulsen Books

APPENDIX A: IN-PERSON AND TELEPHONE PLEA

In-Person and Telephone Plea

Hi [PROSPECTIVE PARTICIPANT],

As you may know, I am a doctoral candidate in Montclair State University's Family Science and Human Development program. I am conducting a research study about our family. You are eligible to participate in this study because you are my [great-aunt, uncle, mother, sibling, spouse, child].

This study will require an in-person interview. After the interview, you will be invited to review your interview transcript and will be allowed to make modifications to the transcribed conversation. Participants will receive a \$25 gift card as a token of appreciation for their participation.

It will take about 1-2 hours of your time. If you have any questions about this study, please contact Dr. Jennifer Urban at 973-655-6884 or email her at <u>urbanj@montclair.edu</u>.

Thank you for considering participation in this study. This study has been approved by the Montclair State University Institutional Review Board, Study no. IRB-FY19-20-1590.

If you are interested in participating, could I please have your email address or phone number so that I can send you a copy of the consent form? I will send you a link to the consent form, which has more information about the study.

APPENDIX B: INTERVIEW PROTOCOLS

Swann-Jackson Adult Interview Protocol – For Generation 1 (G1)

Today is [DATE]. This is the start of an interview with [PARTICIPANT] at [HIS/HER] home at [ADDRESS] in [CITY/STATE]. My name is Rebecca Swann-Jackson and I'll be the interviewer. I am [RELATIONSHIP to PARTICIPANT] and this interview is being done in connection with the history of the Swann and Sisco families. We'll mainly be talking about my [PARTICIPANT'S RELATIONSHIP TO ME]'s recollection of our family experiences since [HIS/HER] childhood.

Tell me who you are.

How do we know each other?

Tell me about your childhood.

How would you describe [PARTICIPANT'S HOMETOWN] as a place to grow up in during the 1940s?

What was your childhood experience in this community?

I never met your grandparents. Tell me about them.

- What did they do?
- What were they like as a couple?
- What were they like with your parents?
- What were they like with you?
- Tell me a story about them.

I never met your parents either. Tell me about them.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- Tell me a story about them.

What was an average day with your parents like?

My grandmother was your sister.

- What was she like as a young woman?
- What was she like as a mother?
- What were her goals in life?

Tell me about my grandparents from your perspective.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- What were they like as parents?

[Show photo of Priscilla Sisco Swann at the White House]

- What do you know about this?
- What are your memories of this event?

Tell me about my mom and dad from your perspective.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- What were they like as parents?

Thinking about yourself as a parent, what about the way you were a parent to your kids is similar to the way your parents were with you? What is different?

Are there things about me that remind you of your grandparents, parents, or my mom and dad? What are they?

Swann-Jackson Adult Interview Protocol – For Generation 2 (G2)

Today is [DATE]. This is the start of an interview with [PARTICIPANT] at [HIS/HER] home at [ADDRESS] in [CITY/STATE]. My name is Rebecca Swann-Jackson and I'll be the interviewer. I am [RELATIONSHIP to PARTICIPANT] and this interview is being done in connection with the history of the Swann and Sisco families. We'll mainly be talking about my [PARTICIPANT'S RELATIONSHIP TO ME]'s recollection of our family experiences since [HIS/HER] childhood.

Tell me who you are.

How do we know each other?

How would you describe [PARTICIPANT'S HOMETOWN] as a place to grow up in during the 1960s?

What was your childhood experience in this community?

I never met your grandparents. Tell me about them.

- What were they like as a couple?
- What were they like with your parents?
- What were they like with you?
- Tell me a story about them.

Tell me about your parents.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- Tell me a story about them.

What was an average day with your parents like?

[Show photo of Priscilla Sisco Swann at the White House]

- What do you know about this?
- What are your memories of this event?

Tell me about my mom and dad from your perspective.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- What were they like as parents?

Thinking about yourself as a parent, what about the way you were a parent to your kids is similar to the way your parents were with you? What is different?

Are there things about me that remind you of your grandparents, parents, or my mom and dad? What are they?

Swann-Jackson Adult Interview Protocol – For Generation 3 (G3)

Today is [DATE]. This is the start of an interview with [PARTICIPANT] at [HIS/HER] home at [ADDRESS] in [CITY/STATE]. My name is Rebecca Swann-Jackson and I'll be the interviewer. I am [RELATIONSHIP to PARTICIPANT] and this interview is being done in connection with the history of the Swann and Sisco families. We'll mainly be talking about my [PARTICIPANT'S RELATIONSHIP TO ME]'s recollection of our family experiences since [HIS/HER] childhood.

Tell me who you are.

How do we know each other?

Tell me about your childhood.

How would you describe [PARTICIPANT'S HOMETOWN] as a place to grow up in during the 1980s?

What was your childhood experience in this community?

Tell me about your Swann grandparents. What did they do?

- What were they like as a couple?
- What were they like with your parents?
- What were they like with you?
- Tell me a story about them.

Tell me about your parents.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- Tell me a story about them.

What was an average day with your parents like?

[Show photo of Priscilla Sisco Swann at the White House]

- What do you know about this?
- What are your memories of this event?

Tell me about my mom and dad from your perspective.

- What did they do?
- What were they like as a couple?
- What were they like with you?
- What were they like as parents?

Thinking about yourself as a parent, what about the way you were a parent to your kids is similar to the way your parents were with you? What is different?

Are there things about me that remind you of your grandparents, parents, or my mom and dad? What are they?

Swann-Jackson Adult Interview Protocol – For Generation 4 (G4)

Today is [DATE]. This is the start of an interview with [PARTICIPANT] at [HIS/HER] home at [ADDRESS] in [CITY/STATE]. My name is Rebecca Swann-Jackson and I'll be the interviewer. I am [RELATIONSHIP to PARTICIPANT] and this interview is being done in connection with the history of the Swann and Sisco families. We'll mainly be talking about my [PARTICIPANT'S RELATIONSHIP TO ME]'s recollection of our family experiences since [HIS/HER] childhood.

Tell me who you are.

How do we know each other?

Tell me about your childhood.

What is it like in your town?

What is an average day with your parents like?

[Show photo of Priscilla Sisco Swann at the White House] What do you know about this?

Tell me about your mom and dad.

- What do they do?
- What are they like as a couple?
- What are they like with you?
- What are they like as parents?

What do you think you'll be like when/if you are married? Why do you think so?

What things do you think will be important to you when/if you are married?

What do you think you'll be like when/if you're a dad? Why do you think so?

What things do you think will be important to you when/if you're a dad?

APPENDIX C: ADULT CONSENT FORM



College of Education and Human Services Department of Family Science and Human

Voice: 973-655-4171 Fax: 973-655-6795

Development

ADULT CONSENT FORM

Please read below with care. You can ask questions at any time, now or later. You can talk to other people before you sign this form.

<u>Title:</u> Using Memory as Method: Studying the Intergenerational Transmission of Egalitarianism in my Black Family

Study Number: IRB-FY19-20-1590

<u>Who am I?</u> I am Rebecca Swann-Jackson. I am a student at Montclair State University in the Family Science and Human Development department.

<u>Why is this study being done?</u> I want to learn about whether or not people in our family work together. If they do, I want to know how. I will be doing interviews with our family members. I also will be looking at historical documents.

<u>What will happen while you are in the study?</u> If you choose to be in this study, I will interview you. This means, I will ask you questions about our family. I want to know what you have seen our family do and how that influenced the things you do or not.

The first thing you have to do is read this paper. Then, you will decide if you want to participate or not. If you agree to participate, I would like tape your interview with an electronic recorder. If you agree, I will turn on the recorder and ask you questions. You can ask me questions too. After the interview, I will transcribe the interview. This means that I will type up all the words you and I said so that I can read them later. This is called a transcript. When I am done with all the interviews, I will read all the transcripts. Then I will figure out the things that were similar or different from all the transcripts. After this study is done, I will delete the interviews and transcripts.

Time: Your interview will take about 1-2 hours.

<u>Risks:</u> You might feel sad when we talk about family members who have passed away. You also might feel bored because you think I know the answers already. You can ask to stop the interview at any time. At a later date, you can ask me not to include things you've told me in the interview. The risks are no greater than those in ordinary life.

Benefits: There are no benefits to you being in this study but I want to hear all of your ideas. Other people who read my study might benefit because they will learn more about how families work.

Name of Faculty Sponsor

<u>Compensation</u>: To compensate you for the time you spend in this study, you will receive a \$25 gift card. You will receive this compensation even if you stop the interview or withdraw from this study.

Who will know that you are in this study? Your name will not be linked to any presentations or reports. I will keep who you are confidential. When I write about my study, I will talk about you but I will not mention you by name. We will make up a secret code word to disguise who you are, but someone might be able to figure it out.

You should know that New Jersey requires that any person having reasonable cause to believe that a child has been subjected to child abuse or acts of child abuse shall report the same immediately to the Division of Youth and Family Services.

<u>Do you have to be in the study?</u> You do not have to be in this study. You are a volunteer! It is okay if you want to stop at any time and not be included in the study. You do not have to answer any questions you do not want to answer.

<u>Do you have any questions about this study?</u> If you have any questions about this study, call Dr. Jennifer Urban at <u>973-655-6884</u> or email her at <u>urbanj@montclair.edu</u>.

Do you have any questions about your rights as a research participant? Call or email the

IRB Chair, Dr. Dana Levitt, at 973-65	55-2097 or reviewboard@montclair.	<u>edu</u> .
It is okay to use my data in other stud	ies: Please initial:	Yes No
It is okay to audiotape me while I am	in this study: Please initial:	Yes No
One copy of this consent form is for	you to keep.	
Statement of Consent		
I have read this form and decided that purposes, the particulars of involvement explained to my satisfaction. I unders indicates that I am 18 years of age or	ent, and possible risks and inconveni tand that I can withdraw at any time.	ences have been . My signature also
Print your name here	Sign your name here	Date
Rebecca Swann-Jackson		
Name of Principal Investigator	Signature	Date
Dr. Jennifer Urban		

Signature

Date

APPENDIX D: PARENT/GUARDIAN CONSENT FORM



PARENT/GUARDIAN CONSENT FORM

Please read below with care. You can ask questions at any time, now or later. You can talk to other people before you fill in this form.

<u>Title:</u> Using Memory as Method: Studying the Intergenerational Transmission of Egalitarianism in my Black Family

Study Number: IRB-FY19-20-1590

Who am I? I am Rebecca Swann-Jackson. I am a student at Montclair State University in the Family Science and Human Development department.

Why is this study being done? I want to learn about whether or not people in our family work together. If they do, I want to know how. I will be doing interviews with our family members. I also will be looking at historical documents.

What will happen while your child or dependent is in the study? If your child chooses to be in this study, I will interview him. This means, I will ask him questions about our family. I want to know what he has seen our family do. I want to know what he might do when he is older.

The first thing your child will do is read this paper. Then, your child will decide if he wants to participate or not. If your child agrees to participate, I would like to audio record the interview with an electronic recorder. If your child agrees, I will turn on the recorder and ask him questions. He can ask me questions too. After the interview, I will transcribe the interview. This means that I will type up all the words your child and I said so that I can read them later. When I am done with all the interviews, I will read all the transcripts. Then I will figure out the things that were similar or different from all the transcripts. After this study is done, I will delete the interviews and transcripts.

Time: Your child's interview will take about 1 hour.

<u>Risks:</u> Your child or dependent may feel sad when we talk about family members who have passed away. Your child also might feel bored because he might think I know the answers already. Your child can ask to stop the interview at any time. At a later date, your child can ask me not to include things he's told me in the interview. The risks are no greater than those in ordinary life.

Benefits: There are no benefits to your child being in this study but I want to hear all of your child's ideas. Other people who read my study might benefit because they will learn more about how families work.

<u>Compensation</u>: To compensate your child for the time he spends in this study, he will receive a \$25 gift card. He will receive the gift card even if he stops the interview or tells me that he doesn't want to participate anymore.

Name of Principal Investigator

Who will know that your child or dependent is in this study? You and our family will know that your child might be in this study.

Your child's name will not be linked to any presentations or reports. I will keep who he is confidential. When I write about my study, I will talk about him but I will not mention him by name. Your child and I will make up a secret code word to disguise who he is, but someone might be able to figure it out.

You should know that New Jersey requires that any person having reasonable cause to believe that a child has been subjected to child abuse or acts of child abuse shall report the same immediately to the Division of Youth and Family Services.

Does your child or dependent have to be in the study?

Your child or dependent does not have to be in this study. He is a volunteer! It is okay if he wants to stop at any time and not be in the study. He does not have to answer any questions that he does not want to answer. Nothing will happen to your child or dependent. He will still get the things that were promised.

<u>Do you have any questions about this study?</u> If you have any questions about this study, call Dr. Jennifer Urban at <u>973-655-6884</u> or email her at <u>urbanj@montclair.edu</u>.

Do you have any questions about you IRB Chair, Dr. Dana Levitt, at 973-655			mail the
	_		
It is okay to use my child's data in other	er studies: Please initial:	Yes	No
It is okay to audiotape my child while l	ne is in this study: Please initial:	Yes	No
One copy of this consent form is for	you to keep.		
Statement of Consent I have read this form and decided that I above. Its general purposes, the particular inconveniences have been explained to at any time. My signature also indicate If you choose to have your child or dep	lars of involvement, and possible rise my satisfaction. I understand that not start I have received a copy of this	sks and ny child car consent for	n withdraw rm.
Child's Name:			
Name of Parent/Guardian	Signature	Da	te
Name of Parent/Guardian	Signature	Da	te

Signature

Date

APPENDIX E: ASSENT FORM



College of Education and Human Services

Department of Family Science and Human Development

Voice: 973-655-4171 Fax: 973-655-6795

ASSENT FORM

Please read below with care. You can ask questions at any time, now or later. You can talk to other people before you fill in this form.

<u>Who am I?</u> I am Rebecca Swann-Jackson. I am a student at Montclair State University in the Family Science and Human Development department.

<u>Why is this study being done?</u> I want to learn about whether or not people in our family work together. If they do, I want to know how. I will be doing interviews with our family members. I also will be looking at other old photos (from before you were born.)

<u>What will happen while you are in the study?</u> I will interview you if you choose to be in this study. This means, I will ask you questions about our family. I want to know what you have seen our family do. I want to learn about what you might do when you are older.

The first thing you have to do is read this paper. Then, you will decide if you want to participate or not. I would like tape your interview with an electronic recorder if you participate. I will turn on the recorder if you say it's okay. Then I will ask you questions. You can ask me questions too. After the interview, I will transcribe the interview. This means that I will type up all the words you and I say so that I can read them later. This is called a transcript. I will read all the transcripts when I am done with all the interviews. I will figure out the things that were the same or different from all the transcripts. I will delete the interviews and transcripts after this study is done.

<u>Time:</u> Your interview will take about 1 hour.

<u>Risks:</u> You might feel sad when we talk about family members who died. You also might feel bored because you think I know the answers already. You can ask to stop the interview at any time. Later, you can ask me not to include things you've told me in the interview.

<u>Benefits:</u> I want to hear all of your ideas. Other people who read my study might learn more about how families work. There are no benefits to you personally.

<u>Compensation</u>: You will receive a \$25 gift card. You will receive the gift card even if you stop the interview or tell me that you don't want to participate anymore.

<u>Who will know that you might be in this study?</u> You and your parents will know that you are in this study. We will make up a secret code word to hide who you are. Someone might be able to figure it out. I will talk about you when I write about my study. I will not say your name. I will only use the code word.

<u>Do you have to be in the study?</u> You do not have to be in this study. I won't get mad with you if you say no. It is okay if you change your mind at any time and leave the study. You do not have to answer any questions you do not want to answer. Nothing will happen to you. You will still get the things that you were promised.

<u>Do you have any questions about this study?</u> If you have any questions about this study, call Dr. Jennifer Urban at <u>973-655-6884</u> or email her at <u>urbanj@montclair.edu</u>.

<u>Do you have any questions about your rights as a research participant?</u> Call or email the IRB Chair, Dr. Dana Levitt, at 973-655-2097 or <u>reviewboard@montclair.edu</u>.

It is okay to use my data in other studie	es:	
Please initial: Yes	No	
It is okay to (audiotape, videotape, or p	photograph) me while I am in this s	study:
Please initial: Yes	No	
Print your name here	Sign your name here	Date
Name of Witness	 Signature	
	g	
Rebecca Swann-Jackson		<u> </u>
Name of Principal Investigator	Signature	Date
Dr. Jennifer Urban		
Name of Faculty Sponsor	Signature	Date

This page intentionally left blank.